

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Barbie, Martin; Hagedorn, Marcus; Kaul, Ashok

Working Paper Dynamic Efficiency and Pareto Optimality in a Stochastic OLG Model with Production and Social Security

IZA Discussion Papers, No. 209

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Barbie, Martin; Hagedorn, Marcus; Kaul, Ashok (2000) : Dynamic Efficiency and Pareto Optimality in a Stochastic OLG Model with Production and Social Security, IZA Discussion Papers, No. 209, Institute for the Study of Labor (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/21054

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

IZA DP No. 209

Dynamic Efficiency and Pareto Optimality in a Stochastic OLG Model with Production and Social Security

Martin Barbie Marcus Hagedorn Ashok Kaul

October 2000

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Dynamic Efficiency and Pareto Optimality in a Stochastic OLG Model with Production and Social Security

Martin Barbie

Department of Economics, University of Bonn

Marcus Hagedorn

IZA, Bonn

Ashok Kaul

IZA, Bonn

Discussion Paper No. 209 October 2000

IZA

P.O. Box 7240 D-53072 Bonn Germany

Tel.: +49-228-3894-0 Fax: +49-228-3894-210 Email: iza@iza.org

This Discussion Paper is issued within the framework of IZA's research area *The Welfare State and Labor Markets*. Any opinions expressed here are those of the author(s) and not those of the institute. Research disseminated by IZA may include views on policy, but the institute itself takes no institutional policy positions.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent, nonprofit limited liability company (Gesellschaft mit beschränkter Haftung) supported by the Deutsche Post AG. The center is associated with the University of Bonn and offers a stimulating research environment through its research networks, research support, and visitors and doctoral programs. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public. The current research program deals with (1) mobility and flexibility of labor markets, (2) internationalization of labor markets in transition, (5) the future of work, (6) project evaluation and (7) general labor economics.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character.

IZA Discussion Paper No. 209 October 2000

ABSTRACT

Dynamic Efficiency and Pareto Optimality in a Stochastic OLG Model with Production and Social Security^{*}

We analyze the interaction between risk sharing and capital accumulation in a stochastic OLG model with production. We give a complete characterization of interim Pareto optimality. Our characterization also subsumes equilibria with a PAYG social security system. In a competitive equilibrium interim Pareto optimality is equivalent to intergenerational exchange efficiency, which in turn implies dynamic efficiency. Furthermore, contrary to the case of certainty, dynamic efficiency does not rule out a Pareto-improving role for a social security system. Social security can provide insurance against macroeconomic risk, namely aggregate productivity risk in the second period of life (old age) through dynamic risk sharing. The mechanism through which social security can Pareto-improve market allocations resembles a Ponzi scheme. But instead of rolling over debt, we can interpret our scheme as one that raises contributions and then rolls over an insurance contract.

JEL Classification: D61, H55

Keywords: Stochastic OLG Model, Dynamic Efficiency, Interim Pareto Optimality, Social Security, Risk Sharing

Ashok Kaul IZA P.O. Box 7240 D-53072 Bonn Germany Tel.: +49 228 38 94 508 Fax: +49 228 38 94 510 Email: kaul@iza.org

^{*} Barbie: Department of Economics, University of Bonn, Adenauerallee 24-42, D-53 3 Bonn, Ger-many. martin.barbie@wiwi.uni-bonn.de; Hagedorn and Kaul: Institute for the Study of Labor (IZA), P.O. Box, 7240, D-53072 Bonn, Germany. hagedorn@iza.org, kaul@iza.org. We are indebted to Georg Nöldeke and in particular to Heraklis Polemarchakis for many helpful comments. Furthermore, we thank Don Brown, John Geanakoplos and seminar participants at Bonn, Louvain-la-Neuve, Yale and the European Workshop for General Equilibrium Theory 2000 in Paris for helpful discussions.

1 Introduction

Social security old age insurance has been a policy issue for decades in all countries with pay-as-you-go public pension systems. A key policy question is whether it is better to finance social security on a pay-as-you-go (PAYG) or fully funded basis. Until recently, academics and policymakers have mostly used deterministic models to analyze this fundamental question. Recent proposals to privatize social security in the US, in addition to less radical proposals such as the social security administration's plan to invest a portion of the Social Security Trust Fund in equities, have highlighted that the deficient treatment of risk and uncertainty might seriously flaw policy analysis. The reason for this flaw is that many proposed policy reforms involve subtle changes in who bears various risks. It has only been in the last few years that theoretical and numerical models have incorporated uncertainty in order to analyze these issues. This has lead to a renewed interest in overlapping generations models.

Most of the theoretical literature on social security is based on the Diamond (1965) OLG model with capital. Under *certainty*, the question of whether to finance social security on a fully funded basis or whether there is a Pareto-improving role for a PAYG system then translates to the question of dynamic efficiency of competitive equilibria (Bose (1974), Breyer (1989), Bose and Ray (1993)).

Under *uncertainty*, the analysis of social security becomes conceptually more complex. The reason for this complexity is that, apart from the issue of efficient capital accumulation, risk sharing issues have to be considered. We analyze the interaction between risk sharing and capital accumulation in a stochastic OLG model with production. This allows us to derive implications about the Pareto optimality of competitive equilibria with a redistributive transfer scheme like a PAYG social security system. We characterize conditions under which the dynamic risk sharing opportunities of a PAYG social security system may lead to a Pareto-improvement of a pure market allocation.

In the development of the OLG literature three themes concerning efficiency can be identified: efficient intergenerational exchange, efficient production (overaccumulation of capital) and Pareto optimality. The *first theme* of efficient intergenerational exchange was already mentioned in the seminal paper by Samuelson (1958). As is well known, in pure exchange OLG economies the first welfare theorem may fail to hold, i.e. competitive equilibria may fail to be Pareto optimal. The contributions to this theme are concerned with the reasons for this failure. A characterization of efficient exchange was given by Balasko and Shell (1980) and Okuno and Zilcha (1980) in a pure exchange OLG model under certainty. The second theme was introduced in OLG models in the celebrated contribution by Diamond (1965) where the question of overaccumulation of capital (dynamic efficiency) is examined. The first complete characterization of dynamic efficiency was given in the context of an infinite horizon production model by Cass (1972). Tirole (1985) has analyzed the relationship between dynamic efficiency and the existence of bubbles as well as the Pareto optimality of bubbly equilibria in an OLG model with production under certainty. An extension of the dynamic efficiency issue under certainty to a setting with uncertainty has been given by Zilcha (1990) and Dechert and Yamamoto (1992). They derived complete characterizations of dynamic efficiency in stochastic OLG models. However, they do not deal with risk sharing issues and hence Pareto optimality.

A third theme that has received considerable interest is that of Pareto optimality of equilibria in OLG models. As can be seen from the discussion above, this question is closely related to the other two themes. *First*, in the pure exchange case under certainty, the question of exchange efficiency (theme 1) and that of Pareto optimality (theme 3) are obviously equivalent. In fact, the characterizations given in Balasko and Shell (1980) and Okuno and Zilcha (1980) are stated in terms of Pareto optimality. Our terminology, exchange efficiency, is introduced to highlight that Pareto optimality in a general setting (with production) consists of three distinct issues: efficient exchange, efficient production and impossibility of improving by joint changes in distribution and production of commodities. *Second*, under certainty it can be shown that efficient production (theme 2) already implies Pareto optimality (theme 3) in a competitive equilibrium (see Bose and Ray (1993) for a discussion). Under uncertainty, however, the relationship becomes more complex because risk sharing issues have to be considered.

At this point we have to be more precise about the notion of Pareto optimality adopted under uncertainty. The definition most often used in the literature is that of *interim Pareto optimality*.¹ This concept considers agents born in different states as distinct agents. Then the usual concept of Pareto optimality is applied to this set of agents. We also make use of this concept of optimality.²

The use of interim Pareto optimality excludes Pareto-improvements through risk sharing that arise from the market incompleteness implied by the missing insurance possibil-

¹The terms interim and conditional Pareto optimality are not used consistently in the literature. We follow Demange and Laroque (1999) and use interim Pareto optimality for allocations which are optimal among *all* feasible allocations, not only stationary ones.

²An alternative notion of optimality would be ex ante Pareto optimality. There, agents born in different states are not considered distinct agents. Instead of that, utility is evaluated before an agent is born, i.e. in expected terms. A good discussion of alternative concepts of Pareto optimality can be found in Dutta and Polemarchakis (1990) and also in Chattopadhyay and Gottardi (1999).

ities against the state in which an individual is born. If it is assumed that markets are complete once an individual is born (sequentially complete markets), then the remaining risk sharing possibilities (if any) follow from the dynamic structure of the economy. We therefore call this *dynamic risk sharing*. It may be possible to introduce a kind of intertemporal insurance which works as an intergenerational transfer under certainty with the exception that transfers in the second period of life may be different in distinct states of the world and thus also incorporate an insurance aspect. A first result concerning interim Pareto optimality was obtained in an influential paper by Abel, Mankiw, Summers and Zeckhauser (1989). They derive a strong sufficient condition for interim Pareto optimality of a market equilibrium. A characterization of exchange efficiency in a pure exchange model under uncertainty was derived by Chattopadhyay and Gottardi (1999), extending the result by Balasko and Shell (1980). As under certainty, exchange efficiency is of course equivalent to interim Pareto optimality in a pure exchange setting under uncertainty.

Our first main result gives a complete characterization of interim Pareto optimality in a stochastic OLG model with production. It turns out that the concept of interim Pareto optimality is equivalent to exchange efficiency in a competitive equilibrium. Furthermore, exchange efficiency implies dynamic efficiency in a competitive equilibrium. In particular this means that extending a pure exchange model to production does not improve the possibilities of risk sharing, although the redistributional possibilities improve in comparison to a pure exchange model. This implies that under exchange efficiency, there exists no pure redistributive transfer system like a PAYG social security system that is Pareto improving. Our second main result shows that, contrary to the case of certainty, the conditions for dynamic efficiency and exchange efficiency do not coincide under uncertainty. Our analysis shows that under the interim Pareto optimality concept the important efficiency benchmark is exchange efficiency. This means that the possibility of overaccumulation of capital is not necessarily related to the risk sharing part of the efficiency problem. As argued above, under certainty, dynamic efficiency implies exchange efficiency, hence under certainty the concepts of exchange efficiency, dynamic efficiency and interim Pareto optimality are all equivalent in a competitive equilibrium. Under uncertainty however, dynamic efficiency is weaker than interim Pareto optimality. This means that dynamic efficiency does not rule out dynamic risk sharing possibilities which could be implemented by a PAYG social security system. A PAYG system can thus offer insurance against aggregate productivity risk in the second period of life (old age).

We can interpret the mechanism through which social security can Pareto-improve

market allocations as a *sophisticated Ponzi scheme*. It closely resembles a Ponzi scheme. But instead of rolling over debt, we can interpret our scheme as one that collects contributions and then rolls over an insurance contract in exchange for the contributions. Our results show that dynamic efficiency does not rule out this kind of dynamic risk sharing.

The characterization does not only answer the question whether a market equilibrium without social security is suboptimal. It also applies to equilibria with social security. This can be seen by noticing that a redistributive policy like social security can replicate monetary equilibria in an OLG model where money is a pure store of value. The fact that the efficiency characterization applies to monetary equilibria as well has been used by Balasko and Shell (1981) in a pure exchange OLG model under certainty and Bose and Ray (1993) in an OLG model with production under certainty. Under uncertainty, Manuelli (1990) examined optimality of monetary equilibria in a pure exchange setup. Aiyagari and Peled (1991) and Demange and Laroque (1999) carry out a similar analysis in an economy with a linear storage technology, where the latter analysis explicitly considers social security equilibria. The contributions under uncertainty, however, restrict attention to stationary allocations, whereas our analysis (with neoclassical production technology and uncertainty) does not require this.

The paper is organized as follows. In section 2 the model is developed. In section 3, we introduce the notions of exchange efficiency and dynamic efficiency in our setting. Section 4 features our first main result, the characterization of interim Pareto optimality in a stochastic OLG model with production (and social security). Section 5 presents a Second Welfare Theorem for our economy and discusses the Pareto-improving role for a PAYG social security system. Section 6 presents our second main result, an example in which dynamic efficiency and interim Pareto optimality do not coincide. Section 7 gives a sufficient condition for interim Pareto optimality and relates the condition to the existence of land. Section 8 concludes.

2 The Model

We consider a stochastic version of the Diamond model (Diamond (1965)). Uncertainty enters the model via shocks to the production technology. Time is discrete, starts at 0 and extends infinitely into the future. There is production and a consumption/saving decision at every point of time. The production technology at time t is described by a function $F : \mathbb{R}^2_+ \times A_t \to \mathbb{R}_+$ where $F(K_t, L_t, \theta_t)$ is the output produced at time t given the capital stock is K_t , labor input is L_t and the current stochastic shock is θ_t . The perishable good produced by the technology is the only good in the economy and is used for production and consumption. There is one representative consumer born per period of time and state of the economy who inelastically supplies one unit of labor in his youth and lives for two periods and trades the consumption good on sequentially complete markets. Due to the production shocks, he has an uncertain second period of life. There is no population growth. This assumption is only made for the simplicity of exposition. It can easily, at the cost of some additional notation, be dispensed with. Further, again for simplicity, the depreciation rate rate is assumed to be 1.

More specifically, the production function satisfies:

• $F(K_t, L_t, \theta_t)$ is homogenous of degree 1 in K_t, L_t , strictly increasing, strictly concave and twice continuously differentiable in K_t, L_t . Further $F(0, L_t, \theta_t) = F(K_t, 0, \theta_t) =$ 0. It also satisfies the Inada conditions $\lim_{K_t \to 0} F_K(K_t, L_t, \theta_t) = \infty$ and

 $\lim_{K_t\to\infty} F_K(K_t, L_t, \theta_t) = 0.$ As usual, define $f(k_t, \theta_t) = F\left(\frac{K_t}{L_t}, 1, \theta_t\right)$, the per capita production function. It inherits from F the following properties:

$$f' = \frac{\partial f}{\partial k} > 0, \ f'' < 0, \ f'(0,\theta_t) = \infty, \ f'(\infty,\theta_t) = 0.$$

For each period in time t, the set of production shocks, A_t , is assumed to have finite cardinality with A_0 being single valued. Consider all sequences of the form $(\theta_0, \theta_1, \theta_2, ...)$ where $\theta_i \in A_i$. These sequences form an uncountable set, denoted Ω . Let \mathcal{A} denote the σ -algebra generated by the product topology on Ω , if each A_i is endowed with the discrete topology. Let $(\Omega, \mathcal{A}, \mu)$ be a measure space with probability measure μ which is assumed to satisfy:

If $\{\theta_0\} \times \{\theta_1\} \times ... \times \{\theta_t\} \times \{\theta_{t+1}\} \times A_{t+2} \times ... \in \mathcal{A}$ is given, then the conditional probability $q_{t+1}(\theta_{t+1} | \sigma_t) = \frac{\mu(\{\theta_0\} \times \{\theta_1\} \times ... \times \{\theta_t\} \times \{\theta_{t+1}\} \times A_{t+2} \times ...)}{\mu(\{\theta_0\} \times \{\theta_1\} \times ... \times \{\theta_t\} \times A_{t+1} \times ...)}$ with $\sigma_t = (\theta_0, \theta_1, \theta_2, ..., \theta_t)$ is well defined and strictly positive for every $\theta_{t+1} \in A_{t+1}$.

Given that the production shocks are the only source of uncertainty in the economy, it is possible to describe the uncertainty by a date-event tree, where $\sigma_0 = \{\theta_0\}$ is the root, $\sigma_t = (\theta_0, \theta_1, \theta_2, ..., \theta_t)$ is a node at time t. The set of nodes at time t is therefore $A_0 \times A_1 \times A_2 \times ... \times A_t$ and denoted by Σ_t . The date-event tree Γ is therefore equal to $\Sigma = \bigcup_{t\geq 0} \Sigma_t$. The generic element of Γ will be denoted by σ . Further we can define the functions $g^t : \Sigma_{t+1} \to \Sigma_t$ by $g^t(\theta_0, \theta_1, \theta_2, ..., \theta_t, \theta_{t+1}) = (\theta_0, \theta_1, \theta_2, ..., \theta_t)$, i.e. g^t assigns to each node in Σ_{t+1} its predecessor in Σ_t . The unique predecessor of a node σ will also be denoted by σ_{-1} . σ^+ denotes the set of nodes which are successors of node σ , i.e. the set of all nodes for which σ is the predecessor. Since the sets A_t are finite, the number of successors of a node is always finite. A path is a sequence of nodes $\{\sigma_t\}$ such that $\sigma_t = g^t(\sigma_{t+1})$ and a generic path will be denoted by σ^{∞} . $t(\sigma)$ denotes the period of time at which event $\sigma \in \Sigma_t$ occurs.

There is one commodity available at each node in the tree and one consumer is born, who lives for two periods. So agents are here distinguished according to date and state of nature in which they are born. Therefore agents can be identified with the node at which they are born, so that in the rest of the paper the agent born in node σ will be called agent σ .

The consumption set of agent σ is $\mathbb{R}^{1+S(\sigma)}_+$, where $S(\sigma)$ is the cardinality of A_{t+1} if the agent is born in period t. His preferences will be described by a utility function $u_{\sigma}(x(\sigma;\sigma), (x(\sigma';\sigma))_{\sigma'\in\sigma^+})$. We denote by $x(\sigma) = (x(\sigma;\sigma), (x(\sigma';\sigma))_{\sigma'\in\sigma^+})$ the consumption vector of agent σ , $x(\sigma;\sigma)$ is his consumption in his youth, $(x(\sigma';\sigma))_{\sigma'\in\sigma^+}$ is his consumption in the different states of nature in his old age. In his youth, each agent inelastically supplies one unit of labor. In old age, agents receive interest payments from capital. Let $w(\sigma)$ denote the wage paid in node σ and $R(\sigma)$ the capital income. Throughout the paper assume that $w(\sigma) > 0$ and $R(\sigma) > 0$. The preferences of the agents are assumed to satisfy the following assumptions:

- 1. The agent born in period -1 has preferences which are strictly monotone in the single consumption good in period 0.
- 2. The preferences of other agents are described by a utility function $u_{\sigma} : \mathbb{R}^{1+S(\sigma)}_+ \to \mathbb{R}_+$ which is twice continuously differentiable (in the interior of its domain), strictly increasing and strictly quasiconcave.

The firm's problem is to decide at each node σ how much capital to invest, i.e. after the shock realization in period t. This capital is then used to produce output at the successor nodes of σ . Given the probabilities and prices, the firm tries to maximize expected profits. Let prices $\psi_t(\sigma_t)$ for all $\sigma_t \in \Sigma_t$ and for all $t \ge 0$ be given. Let $k(\sigma_t)$ denote the investment undertaken by the firm in state σ_t . The firm's problem is then

$$\max_{k(\sigma)\geq 0} \sum_{\theta_{t+1}\in A_{t+1}} q_{t+1}\left(\theta_{t+1} | \sigma_t\right) \cdot \psi_{t+1}\left(\theta_{t+1}, \sigma_t\right) \cdot f\left(k\left(\sigma_t\right), \theta_{t+1}\right) - \psi_t\left(\sigma_t\right) \cdot k\left(\sigma_t\right).$$
(1)

Given certain investment decisions $k^*(\sigma)$ unconditional supporting prices can be defined as follows: set $\psi_{-1} \equiv 1$ and define recursively $\psi_t(\sigma_t) \cdot \frac{1}{f'(k(\sigma_t),\theta_{t+1})} = \psi_{t+1}(\sigma')$ for $\sigma' \in \sigma_t^+$. From these supporting prices we derive the contingent claim prices for the sequentially complete markets by setting

$$p(\sigma_t) = \psi_t(\sigma_t) \cdot \prod_{i=0}^t q_i(\theta_i | (\theta_{i-1}, \theta_{i-2}, ..., \theta_0))$$
(2)

for $\sigma_t = (\theta_0, ..., \theta_t)$.

With these contingent prices, by setting $R(\sigma') = p(\sigma') \cdot [f'(k(\sigma), \theta_{t+1}) \cdot k(\sigma)]$ and $w(\sigma) = p(\sigma) \cdot [f(k(g^{t-1}(\sigma)), \theta_t) - f'(k(g^{t-1}(\sigma)), \theta_t) \cdot k(g^{t-1}(\sigma))]$, the consumer problem for consumers born in $t \ge 0$ can be written as

$$\max_{\left(x(\sigma;\sigma),(x(\sigma';\sigma))_{\sigma'\in\sigma^+}\right)\in\mathbb{R}^{1+S(\sigma)}_+} u_{\sigma}\left(x\left(\sigma;\sigma\right),\left(x\left(\sigma';\sigma\right)\right)_{\sigma'\in\sigma^+}\right)$$
(3)

s.t.
$$p(\sigma) \cdot x(\sigma; \sigma) + \sum_{\sigma' \in \sigma^+} p(\sigma') \cdot x(\sigma'; \sigma) \le w(\sigma)$$
.

Next, we define feasible allocations in this economy, the notion of interim Pareto optimality and a competitive equilibrium. For notational convenience allocations will sometimes be simply denoted by (x, k) in the rest of the paper.

Definition 1 A feasible allocation (given initial capital \widetilde{k}) is a tuple (x, k) $= (x (\theta_0, -1), ((x (\sigma; \sigma), (x (\sigma'; \sigma))_{\sigma' \in \sigma^+}))_{\sigma \in \Sigma}) ((k (\sigma))_{\sigma \in \Sigma}, \widetilde{k})) \text{ such that}$ 1. $x (\sigma_0, -1) + x (\sigma_0; \sigma_0) + k (\sigma_0) = f (\widetilde{k}, \theta_0),$ 2. For $\sigma \in \Sigma : x (\sigma'; \sigma) + x (\sigma'; \sigma') + k (\sigma') = f (k (\sigma), \theta) \quad \forall \sigma' = (\theta, \sigma) \in \sigma^+.$

The concept of Pareto optimality for the economy adopted in this paper is now introduced (Muench (1977), Peled (1982)).

Definition 2 A feasible allocation (x, k) is called interim Pareto-optimal if there exists no other feasible allocation $(\widehat{x}, \widehat{k})$ such that $\widehat{x}(\theta_0, -1) \ge x(\theta_0, -1)$ and $u_{\sigma}(\widehat{x}(\sigma; \sigma), (\widehat{x}(\sigma'; \sigma))_{\sigma' \in \sigma^+}) \ge u_{\sigma}(x(\sigma; \sigma), (x(\sigma'; \sigma))_{\sigma' \in \sigma^+})$ for all $\sigma \in \Sigma$, with at least one strict inequality.

Now, we introduce the concept of a competitive equilibrium for the economy. We define a list of contingent prices $(p(\sigma))_{\sigma \in \Sigma}$ as a price system p.

Definition 3 (x^*, k^*, p^*) is a competitive equilibrium if

- 1. (x^*, k^*) is a feasible allocation,
- 2. given the price system p^* and k^* , household σ solves (3),
- 3. given the price system p^* , $k^*(\sigma)$ solves the firm problem (1) $\forall \sigma \in \Sigma$.

Remark 4 Given that the shocks in each period are finite, the tree describing the uncertainty has a countable number of nodes. Existence of a competitive equilibrium in this economy can be proved under suitable conditions on the boundary behavior of the utility function. This can be done by first showing that after a suitable transformation of the single consumer problem with a given youth wage, the competitive prices in such a "static" two period problem exist. The argument can then be extended by induction (see for example Zilcha (1990)).

A convenient and standard assumption we will make is to assume that the upper bound on output $\sup_{\theta \in \bigcup_{t \ge 0} A_t} \{k | f(k, \theta) = k\}$ is finite, so that our economy is bounded, and therefore, all allocations will be bounded above.

3 Dynamic Efficiency and Exchange Efficiency

In the literature on efficiency in OLG models, most models deal only with a characterization of efficiency of exchange economies, i.e. economies without production (Balasko and Shell (1980), Okuno and Zilcha (1980), Geanakoplos and Polemarchakis (1991)). There are, on the other hand, a number of results that deal with characterization of efficiency in infinite production problems, but without considering specific preferences for consumers and thus without explicitly dealing with Pareto optimality (Cass (1972), Benveniste and Gale (1975)). Both results have been extended to uncertainty (Chattopadhyay and Gottardi (1999) extend the former; Zilcha (1990) and Dechert and Yamamoto (1992) the latter). In the rest of the paper, we will call the latter form of efficiency dynamic efficiency and refer to the former as exchange efficiency. Note that under uncertainty, efficient exchange incorporates an efficient allocation of risk, since exchange efficiency is related to consumers' preferences, which in turn reflects their attitudes towards risk. The purpose of this paper is to examine how the two concepts are related to interim Pareto optimality.

To start, we give the definition of dynamic efficiency and note that a sequence of investment decisions is called dynamically inefficient if it is not dynamically efficient.

Definition 5 A sequence of investment decisions $(k(\sigma)_{\sigma\in\Sigma})$ is dynamically efficient (given initial capital \tilde{k}) if there exists no other sequence of investment decisions $(\hat{k}(\sigma)_{\sigma\in\Sigma})$ such

$$f\left(\widehat{k}\left(\sigma\right),\theta\right) - \widehat{k}\left(\sigma'\right) \ge f\left(k\left(\sigma\right),\theta\right) - k\left(\sigma'\right) \qquad \forall \sigma' \in \sigma^{+} \forall \sigma \in \Sigma$$
$$f\left(\widetilde{k},\theta_{1}\right) - \widehat{k}\left(\sigma\right) \ge f\left(\widetilde{k},\theta_{1}\right) - k\left(\sigma\right) \qquad \forall \sigma \in \Sigma_{1}$$

with at least one strict inequality.

Remark 6 In the definition for uncertainty given e.g. in Dechert and Yamamato (1992) the inequalities are only required to hold with probability one with respect to the probability measure μ defined over the set of paths Ω . In the context of interim optimality, however, it seems natural to change the definition in the way mentioned above. Consider e.g. the case in which there is a shock each period according to a continuously distributed random variable. Then the probability of a certain shock occuring at a certain time is zero, i.e. the probability of a certain "node" σ occuring is zero. Nevertheless there is a continuum of consumers, namely those born at successor "nodes" of σ who have this single shock (respectively node σ) in their history. According to the probability-one definition of dynamic efficiency, the production in their life could be zero, so the resulting allocation would generally not be interim Pareto optimal. In this sense the definition in Dechert and Yamamoto (1992) takes the ex ante period zero standpoint.

Dynamic efficiency thus rules out overaccumulation of capital in the sense that a decrease in savings at one or more nodes would allow for a permanently higher consumption level. For dynamic efficiency necessary and sufficient conditions can be derived with the following elasticity conditions on the production function (see Mitra (1979)). They will be used later in the proof of our characterization of interim Pareto optimality.

• There are positive constants m_1, m_2, m_3, m_4 such that for all k > 0 and $\theta \in \bigcup_{t \ge 0} A_t$ the following holds:

$$m_1 \leq \frac{kf'(k,\theta)}{f(k,\theta)} \leq m_2 \quad \text{and} \quad m_3 \leq \frac{-kf^{"}(k,\theta)}{f'(k,\theta)} \leq m_4.$$
 (4)

Before we state a characterization of dynamic efficiency some additional notation is required. Given $\hat{\sigma} \in \Gamma$, we define a *subtree* (of Γ) with root $\hat{\sigma}$, denoted by $\Gamma_{\hat{\sigma}}$, as a collection of nodes such that $\Gamma_{\hat{\sigma}} \subset \Gamma$ and $\Gamma_{\hat{\sigma}}$ is itself a tree with $\hat{\sigma}$ as its root. Given a path σ^{∞} , σ_t^{∞} denotes the *t*-th coordinate of the path. Given an arbitrary subtree $\Gamma_{\hat{\sigma}}$, we

that

define a path in the subtree $\Gamma_{\widehat{\sigma}}$ as a path with the property that for $t \geq t(\widehat{\sigma})$ all the nodes are elements of the subtree and denote it by $\sigma^{\infty}(\Gamma_{\widehat{\sigma}})$, i.e. $\sigma^{\infty}(\Gamma_{\widehat{\sigma}}) \subset \{\sigma_{1}^{\infty}, ..., \sigma_{t(\widehat{\sigma})-1}^{\infty}\} \cup \Gamma_{\widehat{\sigma}}$. For any $\widehat{\sigma} \in \Gamma$, we denote by $\Gamma(\widehat{\sigma}, \Gamma)$ the subtree that has $\widehat{\sigma}$ as its root and includes all successor nodes of $\widehat{\sigma}$.

Under this assumption and the standard assumptions made above, Zilcha (1990) derives the following characterization of dynamic efficiency under uncertainty. The fact that we have slightly changed the definition of dynamic efficiency obviously does not alter the result.

Theorem 7 Under the above assumptions on the production function an interior feasible allocation (x^*, k^*) for which k^* is bounded below, i.e. there exists $\varepsilon > 0$ such that $k^*(\sigma) \ge \varepsilon$ for all $\sigma \in \Gamma$, is dynamically inefficient if and only if there exists a node $\sigma_{\tau} \in \Gamma$ and some C > 0 such that

$$\sum_{t=\tau}^{\infty} \frac{1}{\psi\left(\sigma_{t}\right)} \leq C$$

along every path $\sigma^{\infty} \in \{\sigma_1^{\infty}, ..., \sigma_{\tau-1}^{\infty}\} \cup \Gamma(\sigma_{\tau}, \Gamma)$, where $\psi_t(\sigma_t)$ are the non-contingent supporting prices defined above.³

Let us now illustrate the concept of (intergenerational) exchange efficiency. We can reduce our economy with production to a pure exchange economy by fixing the production and investment/saving decisions of a given competitive equilibrium by taking the now fixed (competitive equilibrium) output as aggregate endowment at each node. If it is not possible to achieve an interim Pareto-improvement by pure transfers of commodities, the allocation is called exchange efficient. More formally, we give the following definition of exchange efficiency in a setting with capital.

Definition 8 An allocation (x^*, k^*) is exchange efficient (given initial capital k) if there is no other feasible allocation (\hat{x}, k^*) which improves upon (x^*, k^*) in the sense of Definition 2.

Note that the fixed level of initial capital stock is not so important here since capital remains fixed anyway. Necessary and sufficient conditions for exchange efficiency are usually given by imposing restrictions on the curvature of indifference surfaces or the production technology. A sufficient condition for exchange efficiency can be derived if the following assumption about the curvature of the utility functions is satisfied.

 $^{^{3}}$ Note that due to the strict monotonicity of preferences and the properties of the technology these prices will be strictly positive.

Definition 9 An interior competitive equilibrium (x^*, k^*, p^*) satisfies the non-vanishing Gaussian curvature condition if there exists a $\underline{\rho} > 0$ such that for all feasible allocations (\hat{x}, k^*)

$$u_{\sigma}(\widehat{x}(\sigma)) \geq u_{\sigma}(x(\sigma))$$

$$\implies \sum_{\sigma' \in \sigma^{+}} \delta_{2}(\sigma', \sigma) \geq -\delta_{1}(\sigma) + \underline{\rho} \frac{(\delta_{1}(\sigma))^{2}}{p^{*}(\sigma)} \quad \forall \sigma \in \Sigma,$$

where $\delta_1(\sigma) = p^*(\sigma) \cdot [\widehat{x}(\sigma; \sigma) - x^*(\sigma; \sigma)]$ and $\delta_2(\sigma', \sigma) = p^*(\sigma') \cdot [\widehat{x}(\sigma'; \sigma) - x^*(\sigma'; \sigma)]$ for $\sigma' \in \sigma^+$. ρ is called the lower curvature coefficient.

Note that in the case without uncertainty, the non-vanishing Gaussian curvature condition imposes a lower bound on the curvature of the indifference curve by approximating it from below by a quadratic polynomial.

Remark 10 The assumption of non-vanishing Gaussian curvature is not very restrictive. This can be seen in many applications. It will be satisfied if the preferences are identical across nodes, satisfy the other assumptions made in this paper and the competitive equilibrium allocation (x^*, k^*) is uniformly bounded away from 0 and alternative allocations (\hat{x}, k^*) are restricted to a neighborhood of (x^*, k^*) : i.e. $||x^*(\sigma) - \hat{x}(\sigma)|| < \kappa$ for some κ sufficiently small, where ||.|| denotes the Euclidian norm. Then our assumptions about preferences imply non-vanishing Gaussian curvature. Note that restricting the allocation (\hat{x}, k^*) to neighborhoods of (x^*, k^*) is without loss of generality when considering whether a Pareto-improvement exists, given the strict quasiconcavity of preferences and the convexity of technology. The same applies for the bounded Gaussian curvature defined next.

Definition 11 An interior competitive equilibrium (x^*, k^*, p^*) satisfies the bounded Gaussian curvature condition if there exists a $\overline{\rho} > 0$ such that for all feasible allocations (\widehat{x}, k^*)

$$p^{*}(\sigma) \cdot [\widehat{x}(\sigma) - x^{*}(\sigma)] < 0 \quad and \quad \sum_{\sigma' \in \sigma^{+}} \delta_{2}(\sigma', \sigma) \ge -\delta_{1}(\sigma) + \overline{\rho} \frac{(\delta_{1}(\sigma))^{2}}{p^{*}(\sigma)}$$

imply $u_{\sigma}\left(\widehat{x}\left(\sigma\right)\right) \geq u_{\sigma}\left(x^{*}\left(\sigma\right)\right),$

where $\delta_1(\sigma) = p^*(\sigma) \cdot [\widehat{x}(\sigma; \sigma) - x^*(\sigma; \sigma)]$ and $\delta_2(\sigma', \sigma) = p^*(\sigma') \cdot [\widehat{x}(\sigma'; \sigma) - x^*(\sigma'; \sigma)]$ for $\sigma' \in \sigma^+$. $\bar{\rho}$ is called the upper curvature coefficient. In order to characterize exchange efficiency we need some more definitions. Given a subtree $\Gamma_{\hat{\sigma}}$, a weight function is a function $\lambda_{\Gamma_{\hat{\sigma}}} : \Gamma_{\hat{\sigma}} \to [0,1]$ such that $\sum_{\sigma' \in \sigma^+ \cap \Gamma_{\hat{\sigma}}} \lambda_{\Gamma_{\hat{\sigma}}} (\sigma') = 1$ for all $\sigma \in \Gamma_{\hat{\sigma}}$. Given a pair $(\Gamma_{\hat{\sigma}}, \lambda_{\Gamma_{\hat{\sigma}}})$ the induced weight function, denoted $\hat{\lambda}_{\Gamma_{\hat{\sigma}}} : \Gamma_{\hat{\sigma}} \times \lambda_{\Gamma_{\hat{\sigma}}} \to [0,1]$, is defined as $\hat{\lambda}_{\Gamma_{\hat{\sigma}}} (\hat{\sigma}) = 1$, $\hat{\lambda}_{\Gamma_{\hat{\sigma}}} (\sigma) = \lambda_{\Gamma_{\hat{\sigma}}} (\sigma) \cdot \hat{\lambda}_{\Gamma_{\hat{\sigma}}} (\sigma_{-1})$ for $\sigma_{-1} \in \Gamma_{\hat{\sigma}}$. Now we can restate a simplified form of a result due to Chattopadhyay and Gottardi (1999).⁴

Theorem 12 Let (x^*, k^*, p^*) be an interior competitive equilibrium which is bounded below, satisfying the non-vanishing Gaussian curvature condition and the bounded Gaussian curvature condition. Then a necessary and sufficient condition for the allocation **not** to be (interim) exchange efficient is that there exists a subtree $\Gamma_{\hat{\sigma}}$, a weight function $\lambda_{\Gamma_{\hat{\sigma}}}$ and a finite number A such that for every path $\sigma^{\infty}(\Gamma_{\hat{\sigma}})$ in the subtree

$$\sum_{t=t(\widehat{\sigma})}^{\infty} \frac{\widehat{\lambda}_{\Gamma_{\widehat{\sigma}}}\left(\sigma_{t}^{\infty}\right)}{p^{*}\left(\sigma_{t}^{\infty}\right)} \leq A$$

4 Characterizing Interim Pareto Optimality

In this section, we present our first main result. It gives a characterization of interim Pareto optimality in a stochastic OLG model with production. Compared to pure exchange, a setting with production allows for the analysis of many real world problems as they often involve capital accumulation. In particular, it allows us to analyze the interplay between capital accumulation and a redistributive transfer scheme like PAYG social security and its risk sharing opportunities. Other applications include the role of government debt or the role of monetary policy as insurance. A priori, it is not clear how the rich additional redistributional possibilities (compared to a pure exchange setting) of an economy with production influence the characterization of interim Pareto optimality.

Theorem 13 Let (x^*, k^*, p^*) be an interior competitive equilibrium which is bounded below. Assume that both the non-vanishing and bounded Gaussian curvature assumptions hold as well as the elasticity assumption on the production function holds. Then a necessary and sufficient condition for the competive equilibrium allocation not to be interim Pareto-optimal is that here exists a subtree $\Gamma_{\hat{\sigma}}$, a weight function $\lambda_{\Gamma_{\hat{\sigma}}}$ and a finite positive

⁴The result in Chattopadhyay and Gottardi (1999) is proved under somewhat weaker assumptions. Under their assumptions, the necessary and sufficient conditions do not coincide (see Theorem 1 and 2 in Chattopadhyay and Gottardi (1999) for the more general statement). We chose our stronger assumptions in order to simplify the exposition.

number A such that for every path $\sigma^{\infty}(\Gamma_{\widehat{\sigma}})$ in the subtree

$$\sum_{t=t(\overline{\sigma})}^{\infty} \frac{\widehat{\lambda}_{\Gamma_{\widehat{\sigma}}}(\sigma_t^{\infty})}{p^*(\sigma_t^{\infty})} \le A.$$
(5)

Therefore exchange efficiency and interim Pareto optimality are equivalent in a competitive equilibrium.

Proof. See the Appendix. ■

The main idea of the proof is to show that a competitive equilibrium is exchange efficient if and only if there is no possibility of Pareto improvement by *joint* feasible deviations (lower consumption *and* investment of the young) of the initial competitive consumption-investment plan.

Remark 14 Our result implies as a special case the equivalence of Pareto optimality, exchange efficiency and dynamic efficiency in an economy without uncertainty under the curvature assumptions on preferences and technology made above (Bose (1975), Bose and Ray (1993)). To see this, note that under certainty the condition in theorem 13 reduces to $\sum_{t=1}^{\infty} \frac{1}{p_t} = \infty$, because the degenerate date-event tree is the only subtree of itself. This condition is equivalent to both exchange and dynamic efficiency (see Balasko and Shell (1980) and Cass (1972)).

Remark 15 It should be noted that the efficiency characterization also carries over to monetary competive equilibria as can easily be seen from the proof. This was also used to characterize the efficiency of monetary equilibria by Balasko and Shell (1981) and Bose and Ray (1993) under certainty.

The characterization in the theorem was derived for an economy without a social security system. The previous remark, however, shows that the theorem also applies to a model where money is a pure store of value. An alternative interpretation of money can be given in terms of a redistributive scheme like PAYG social security (or government debt). The reason for this is that monetary and fiscal policy are equivalent in this class of models. This was already noted by Balasko and Shell (1981). This equivalence carries over to models with uncertainty under the interim Pareto optimality criterion.

In the next section we show that if the condition for Pareto optimality in the theorem is violated in a competitive equilibrium (with or without a social security system) then there is a Pareto improving role for a well designed PAYG social security system.

5 A Second Welfare Theorem

In this section we introduce a transfer scheme which can be interpreted as PAYG social security system. We prove a second welfare theorem for our OLG economy with transfers. Furthermore we show that for every allocation which is not Pareto optimal there exists a feasible Pareto superior and Pareto optimal allocation. These two results together show that a well desingned social security system has a Pareto improving role if the condition for Pareto optimality derived in the previous section is violated.

In exchange OLG models, it is generally possible to prove a version of the second welfare theorem, namely that Pareto optimal allocations can be supported as a competitive equilibrium. In the setup introduced so far this is not possible since our definition of competitive equilibrium with production requires that the young generation earns income only by selling their endowment (one unit of labor) and that the old generation owns the capital. To support any interim Pareto optimal allocation, we must therefore allow for transfers. In the following, $T^*(\sigma; \sigma')$ is a transfer to household σ in state σ' .

Definition 16 A competitive equilibrium with transfers is a tuple (x^*, k^*, p^*, T^*) such that

- 1. (x^*, k^*) is feasible,
- 2. households solve

$$\max_{\left(x(\sigma;\sigma), (x(\sigma';\sigma))_{\sigma'\in\sigma^+}\right)\in\mathbb{R}^{1+S(\sigma)}_+} u_{\sigma}\left(x\left(\sigma;\sigma\right), (x\left(\sigma';\sigma\right))_{\sigma'\in\sigma^+}\right) \tag{6}$$

$$\begin{split} s.t. \quad p\left(\sigma\right) \cdot x\left(\sigma;\sigma\right) + \sum_{\sigma' \in \sigma^+} p\left(\sigma'\right) \cdot x\left(\sigma';\sigma\right) &\leq w\left(\sigma\right) \\ &+ T^*\left(\sigma;\sigma\right) + \sum_{\sigma' \in \sigma^+} T^*\left(\sigma;\sigma'\right), \end{split}$$

3. given the price system p^* , $k^*(\sigma)$ solves the firm problem (1) $\forall \sigma \in \Sigma$,

4. $T^*(\sigma; \sigma) + T^*(\sigma_{-1}; \sigma) = 0 \quad \forall \sigma \in \Sigma.$

Remark 17 The last condition allows us to interpret the introduced transfer scheme as a PAYG social security system. It specifies that in every state σ the transfers to the young $T^*(\sigma; \sigma)$ and the transfers to the old $T^*(\sigma_{-1}; \sigma)$ sum up to zero. Thus our transfer system is balanced in every state of the world.

To prove the second welfare theorem, we want to characterize optimal allocations by first order conditions. Therefore we need some boundary conditions on the utility functions. A precise statement is given in the following remark.

Remark 18 We say that utility functions satisfy boundary conditions if for all $\sigma \in \Sigma$ the corresponding utility functions $u_{\sigma}(.)$ satisfies $\lim_{x(\sigma;\sigma)\to 0} \frac{\partial u_{\sigma}(x(\sigma;\sigma),(x(\sigma';\sigma))_{\sigma'\in\sigma^+})}{\partial x(\sigma;\sigma)} \to \infty$ for all $(x(\sigma';\sigma))_{\sigma'\in\sigma^+} \in \mathbb{R}^{S(\sigma)}_{++}$ and $\lim_{x(\sigma^*;\sigma)\to 0} \frac{\partial u_{\sigma}(x(\sigma;\sigma),(x(\sigma';\sigma))_{\sigma'\in\sigma^+})}{\partial x(\sigma^*;\sigma)} \to \infty$ for $\sigma^* \in \sigma^+$ for all $x(\sigma;\sigma), (x(\sigma';\sigma))_{\sigma'\in\sigma^+\setminus\{\sigma^*\}} \in \mathbb{R}^{S(\sigma)}_{++}$.

Proposition 19 Under boundary conditions on the utility functions, any interior interim Pareto optimal allocation can be supported as a competitive equilibrium with transfers.

Proof. Let (x^*, k^*) be the given interior interim Pareto optimal allocation. If we define an Arrow-Debreu price system by using (2) and construct $\psi(\sigma)$ recursively using k^* from the given allocation, optimality conditions for the firm imply that given these prices k^* is profit maximizing. Give the consumer σ the final income $p(\sigma) \cdot x^*(\sigma; \sigma) + \sum_{\sigma' \in \sigma^+} p(\sigma') \cdot x^*(\sigma'; \sigma)$ by constructing an appropriate transfer scheme. The first order conditions for interim Pareto optimality imply that consumer σ chooses indeed $x^*(\sigma; \sigma), (x^*(\sigma'; \sigma))_{\sigma' \in \sigma^+}$. Since the allocation (x^*, k^*) is by definition also feasible, the result follows.

To apply the second welfare theorem in order to answer the question whether Paretoimproving policies exist, we need the following proposition which shows that for each Pareto suboptimal allocation a Pareto-superior *and* Pareto optimal allocation exists. If this allocation is interior then, by the second welfare theorem, it can be decentralized with appropriate transfers. These transfer can be interpreted as a (non-voluntary) social security system or as non-stationary government debt issued in bonds.⁵

Proposition 20 Let (x, k) be a given feasible allocation. Then there exists another feasible allocation (\tilde{x}, \tilde{k}) such that

- $\left(\widetilde{x},\widetilde{k}\right)$ weakly interim Pareto-dominates (x,k),
- $\left(\widetilde{x},\widetilde{k}\right)$ is interim Pareto optimal.

⁵The resulting equilibria are often called monetary equilibria. The modeling of monetary equilibria under uncertainty is similar to the certainty case in Balasko and Shell (1981). The important difference is that under uncertainty there is not a single Arrow-Debreu price for money. Instead there is a vector $(p^m(\sigma))_{\sigma \in \Sigma}$ which satisfies the arbitrage condition $p^m(\sigma) = \sum_{\sigma' \in \sigma^+} p^m(\sigma')$ for each $\sigma \in \Sigma$.

Proof. The proof is an application of Zorn's lemma. Aliprantis and Border (1994) prove a similar result for a finite economy. By our assumption that the economy is bounded (end of section 2), we can w.l.o.g. assume that the total amount of the single commodity available at each node σ is bounded above by some constant \overline{y} . Let $Y := [0, \overline{y}]^3$ be the threefold product of the $[0, \overline{y}]$ intervall. Define $\mathcal{Y} := \prod_{\sigma \in \Sigma} Y$. By the Tychonoff Product Theorem \mathcal{Y} is compact in the product topology. The set of feasible allocations, which we denote by F, is closed in the product topology and contained in \mathcal{Y} . Therefore it is also compact. Define an equivalence relation \backsim on F by setting $(\widehat{x}, \widehat{k}) \backsim (\widetilde{x}, \widetilde{k}) \iff \widehat{x}(\theta_0, -1) = \widetilde{x}(\theta_0, -1)$ and $u_{\sigma}(\widehat{x}(\sigma; \sigma), (\widehat{x}(\sigma'; \sigma))_{\sigma'\in\sigma^+}) = u_{\sigma}(\widetilde{x}(\sigma; \sigma), (\widetilde{x}(\sigma'; \sigma))_{\sigma'\in\sigma^+})$ for all $\sigma \in \Sigma$. The quotient space \mathcal{Y}/ \backsim endowed with the quotient topology is as a continuous image of a compact set still compact. From now on we identify (x, k) with the corresponding equivalence class. Furthermore, by extending the utility functions in the canonical manner to \mathcal{Y} , they are also continuous on the quotient space \mathcal{Y}/\backsim . Now let the feasible allocation (x,k) be given. Consider the set

$$V_{(x,k)} := \left\{ \left(\widehat{x}, \widehat{k} \right) \in F \left| \widehat{x} \left(\theta_0, -1 \right) \ge x \left(\theta_0, -1 \right) \right. \right\}$$
 and

$$u_{\sigma}\left(\widehat{x}\left(\sigma;\sigma\right),\left(\widehat{x}\left(\sigma';\sigma\right)\right)_{\sigma'\in\sigma^{+}}\right)\geq u_{\sigma}\left(x\left(\sigma;\sigma\right),\left(x\left(\sigma';\sigma\right)\right)_{\sigma'\in\sigma^{+}}\right) \text{ for all } \sigma\in\Sigma\}.$$

Clearly, $V_{(x,k)}$ is non-empty since it contains (x,k). Furthermore, it is closed and therefore compact. Define a partial order \succeq on $V_{(x,k)}$ by setting $(\hat{x}, \hat{k}) \succeq (\tilde{x}, \tilde{k}) \iff$ $V_{(\hat{x}, \hat{k})} \subseteq V_{(\tilde{x}, \tilde{k})}$. Maximal elements with respect to this order are Pareto-optima and are weakly preferred to (x, k) in the order \succeq .

To show that maximal elements exist it suffices by Zorn's lemma to show that every chain has an upper bound. Let $(x_{\alpha}, k_{\alpha}), \alpha \in \Lambda$ be a given chain, where Λ is the index set. Consider the set $V^* := \bigcap_{\alpha \in \Lambda} V_{(x_{\alpha}, k_{\alpha})}$. We have to show that this set is non-empty. Since the $V_{(x_{\alpha}, k_{\alpha})}$ sets are closed subsets of the compact space \mathcal{Y}/\sim , it suffices by the finite intersection property to show that any finite intersection of the $V_{(x_{\alpha}, k_{\alpha})}$ sets is non-empty. But this follows immediately from the fact that the $V_{(x_{\alpha}, k_{\alpha})}$ form a chain. For any (x^*, k^*) we have $V_{(x^*, k^*)} \subseteq V_{(x_{\alpha}, k_{\alpha})}$ for all $\alpha \in \Lambda$ and therefore an upper bound for the chain. Thus the result follows with Zorn's lemma.

The results in this section indicate the scope for Pareto-improvements through a PAYG social security system under uncertainty. If the condition for Pareto optimality derived in the previous section is violated, a well designed social security system can improve the

allocation of risk relative to the pure market outcome (or to some initially given social security system). In the special case of certainty, the corresponding characterization of interim Pareto optimality also indicates a Pareto-improving role for social security. There, however, the role for social security is restricted to simple Ponzi schemes which roll over debt infinitely into the future. The condition for Pareto optimality (equivalently: dynamic efficiency) then exactly rules out this kind of schemes. Under uncertainty, social security may additionally serve as a means of intergenerational risk sharing. The contributions and benefits of the social security system may have to be conditioned on the different realizations of uncertainty to achieve this goal. The crucial point is that these risk sharing opportunities do not result from the fact that markets are incomplete, but from the dynamic structure of the economy. In particular, we do not consider risk sharing against the state in which an individual is born. This kind of risk sharing is ruled out by the choice of our criterion of *interim* Pareto optimality combined with our assumption that markets are complete once an individual is born (sequentially complete markets). Thus risk sharing in our model works through a mechanism that closely resembles a Ponzi scheme, but is more sophisticated. Instead of rolling over debt, we can interpret our scheme as one that raises contributions and then rolls over an insurance contract in exchange for the contributions. Therefore we use the term dynamic risk sharing.

Three things are remarkable about this insurance scheme in the Diamond model under uncertainty. First, the condition under which such an insurance contract is feasible is the same as in a model without production (as in Chattopadhyay and Gottardi (1999)). In particular this means that extending a pure exchange model to production does not improve the possibilities of risk sharing. This is surprising because, in comparison to a pure exchange model, the redistributional possibilities considerably improve by introducing joint (consumption and investment) deviations from a competitive equilibrium as a source of Pareto-improvements.

Second, this kind of social security scheme provides an insurance that cannot be replicated in a capital market. In fact, social security may provide insurance against macroeconomic risk which is often considered to be uninsurable. To be more precise, it provides insurance against the aggregate productivity risk in the second period of life (old age).

Third, contrary to the pure exchange case, we can relate this *theoretical* possibility of insurance against aggregate risk to an *empirically* testable efficiency criterion, namely dynamic efficiency. This is important because real world economies are usually considered to be dynamically efficient (Abel, Summers, Mankiw and Zeckhauser (1989)). Thus simple Ponzi schemes are not a feasible source of Pareto-improvements. The question is then whether the described dynamic risk sharing possibilities, which we interpreted above as sophisticated Ponzi schemes, are relevant in real world economies or whether they are ruled out by the fact that simple Ponzi schemes are not feasible. We answer this question in the next section.

6 Dynamic Efficiency versus Interim Pareto Optimality

In this section we provide our second main result. We demonstrate that under uncertainty dynamic efficiency is not sufficient for interim Pareto optimality in a competitive equilibrium. It is a strictly weaker efficiency benchmark.

Theorem 21 In a stochastic OLG model with production, dynamic efficiency does not rule out (interim) Pareto-improvements in a competitive equilibrium. Thus, even under dynamic efficiency, there may be a Pareto-improving role for a government by introducing a well designed social security system.

Proof. In the following example we construct a generic competitive equilibrium allocation that is dynamically efficient without being exchange efficient respectively interim Pareto optimal. Consider an economy with two possible shocks each period of time. Let $(\alpha_i)_{i=0}^{\infty}$ be a sequence of real numbers $0 < \alpha_i < 1$ with the property $\prod_{i=0}^{\infty} \alpha_i > 0$, which is equivalent to $\sum_{i=0}^{\infty} (1 - \alpha_i) < \infty$. Suppose there is a sequence of shocks $(\theta_t)_{t=0}^{\infty}$ with $\theta_t \in A_t$ for all t such that $\alpha_i = q_i (\theta_i | \theta_{i-1}, ..., \theta_0)$. In other words, there exists a path in the tree which has strictly positive probability. Suppose further that along the path $\sigma^{\infty} = (\theta_0, \theta_1, ...)$ we have $\psi(\sigma_i) = 3^i$ and hence $\sum_{i=0}^{\infty} \frac{1}{\psi(\sigma_i)} < \infty$. Assume furthermore that for every node σ in the tree there exists a path $\tilde{\sigma}^{\infty}$ with $\sigma \in \tilde{\sigma}^{\infty}$ such that $\sum_{i=0}^{\infty} \frac{1}{\psi(\tilde{\sigma}_i)} = \infty$. Clearly the economy described above is **not** dynamically inefficient although the series $\sum_{i=0}^{\infty} \frac{1}{\psi(\sigma_i)}$ converges (and is therefore in this case uniformly bounded) on a set of strictly positive measure, since there is no node in the tree at which a dissaving is possible without a later decrease in consumption.

However, there is a interim Pareto-improving pure redistribution possible. We clearly have $\prod_{i=0}^{t} \alpha_i \cdot \psi(\sigma_t) \ge \left(\prod_{i=0}^{T} \alpha_i\right) \cdot (1-\varepsilon)^{t-T} \cdot \psi(\sigma_t)$ for every $\varepsilon > 0$ and some T when t > T. Since for an ε sufficiently small we have $\sum_{t=T+1}^{\infty} \frac{1}{\frac{(1-\varepsilon)^{t-T}}{\psi(\sigma_t)}} = \sum_{t=T+1}^{\infty} \frac{1}{\frac{(1-\varepsilon)^{t-T}}{3^t}} < \infty$ the claim follows immediately by choosing the degenerate subtree consisting of the path σ^{∞} and by applying our main theorem.

To make the example more concrete, suppose that preferences of the consumers along the convergent path σ^{∞} are of the form $u_{\sigma_t}(x(\sigma)) = \sqrt{x(\sigma;\sigma)} + b(\sigma_t) \cdot \alpha_t \sqrt{x(\sigma_t;\sigma'_{1t})} + b(\sigma_t) \cdot (1 - \alpha_t) \sqrt{x(\sigma_t;\sigma'_{2t})}$, where $b(\sigma_t)$ is a positive real number. Let the technology be given by

$$f\left(k,\theta\right) = a\left(\theta\right) \cdot k^{\beta}$$

where β will be chosen to satisfy a certain condition and $0 < \beta < 1$.

This technology clearly satisfies the elasticity conditions (4). Suppose the individual born in σ_t faces given interest rates of 3 and $\frac{1}{3}$ in the two possible events in his second period of life. The individual's problem is then

$$\max \quad \sqrt{x(\sigma;\sigma)} + b(\sigma_t) \cdot \alpha_t \sqrt{x(\sigma_t;\sigma'_{1t})} + b(\sigma_t) \cdot (1-\alpha_t) \sqrt{x(\sigma_t;\sigma'_{2t})}$$
(7)

s.t.
$$x(\sigma; \sigma) + s(\sigma_t) = w(\sigma_t),$$

 $x(\sigma_t; \sigma'_{1t}) = \frac{1}{3} \cdot s(\sigma_t),$
 $x(\sigma_t; \sigma'_{2t}) = 3 \cdot s(\sigma_t).$

It follows $s(\sigma_t) = \frac{c(\sigma_t)}{1+c(\sigma_t)} \cdot w(\sigma_t)$, where $c(\sigma_t) = (b(\sigma_t))^2 \cdot \left(\alpha_t \sqrt{\frac{1}{3}} + (1-\alpha_t)\sqrt{3}\right)^2$. We want $c(\sigma_t)$ to be independent of the node σ_t and equal an arbitrarily chosen positive real number c. Given the savings decision $s(\sigma_t)$, in order for 3 and $\frac{1}{3}$ to be equilibrium interest rates we must have

$$f'(s(\sigma_t), \theta_1) = a(\theta_1) \cdot \beta \cdot (s(\sigma_t))^{\beta - 1} = \frac{1}{3},$$
(8)

$$f'(s(\sigma_t), \theta_2) = a(\theta_2) \cdot \beta \cdot (s(\sigma_t))^{\beta-1} = 3$$

We know $w(\sigma_t) = (1 - \beta) \cdot a(\theta) \cdot (k(\sigma_{t-1}))^{\beta}$ where $\sigma_t = (\theta, \sigma_{t-1})$. Thus

$$s(\sigma_t) = \frac{c(\sigma_t)}{1 + c(\sigma_t)} \cdot (1 - \beta) \cdot a(\theta) \cdot (k(\sigma_{t-1}))^{\beta}.$$

The function $f(x) = \frac{c}{1+c} \cdot (1-\beta) \cdot a(\theta) \cdot x^{\beta}$ has for fixed $a(\theta)$ a nonzero fixed point, which we call k^* . We can now solve, for an arbitrary given k^* , for the corresponding a, which is given by $a = \frac{1+c}{c} \cdot \frac{1}{1-\beta} \cdot (k^*)^{1-\beta}$. Plugging k^* for $s(\sigma_t)$ and $\frac{1+c}{c} \cdot \frac{1}{1-\beta} \cdot (k^*)^{1-\beta}$ for $a(\theta_1)$ into (8), k^* cancels out and the resulting condition for β is

$$\frac{1+c}{c}\cdot\frac{\beta}{1-\beta}=\frac{1}{3}$$

or equivalently

$$\beta = \frac{\frac{1}{3} \cdot \frac{c}{1+c}}{1 + \frac{1}{3} \cdot \frac{c}{1+c}}.$$

If we choose $a(\theta_2) = 9 \cdot a(\theta_1)$, the second equation above is also satisfied.

Up to now we have constructed a k^* , so that a capital stock of k^* is maintained if along the path σ^{∞} the same shock always occurs. It is clear from our construction that once we deviate from this path, agents have a higher wage income. Hence, if c (which determines the savings behavior together with the wage) remains fixed, households save more. This implies that if the same shocks occur each period, higher capital stock means higher wages and therefore higher savings etc. Thus, as long c remains fixed, the capital stock off the path will never fall below k^* . Furthermore, by $a(\theta_2) \cdot k^{\beta} = k \iff k = a(\theta_2)^{\frac{1}{1-\beta}}$, an upper bound on the maximal possible capital stock is given. Since $a(\theta_1)$ and therefore $a(\theta_2) \cdot \beta \cdot a(\theta_2)^{\frac{\beta}{1-\beta}}$, the lowest possible interest rate when the "high" shock $a(\theta_2)$ occurs, to be strictly larger than 1. Given this lower bound on interest rates, it is possible to determine bounds on $b(\sigma_t)$ if we fix the probabilities of shocks off the path equal to $(\frac{1}{2}, \frac{1}{2})$. On the path we have $\alpha_t \to 1$, and therefore $b(\sigma)$ will converge along the path to a constant.

Overall, the examples display all the features described in the first few paragraphs of this section. The Gaussian curvature assumptions are satisfied since the values of the $b(\sigma_t)$ are in a bounded intervall. This is so because in the expression for $c(\sigma_t)$, under the assumptions made, the value of the second bracket is in an interval, so that to keep $c(\sigma_t)$ constant the $b(\sigma_t)$ can also be chosen from an intervall. Since interest rates are bounded above and below and consumption along σ^{∞} is constant, the overall equilibrium consumption levels are a subset of a compact set. All of this implies the curvature assumptions. In addition, the consumption along the path σ^{∞} is uniformly bounded away from zero.⁶ \blacksquare

The basic idea, why under uncertainty dynamic efficiency is weaker than exchange efficiency and also Pareto optimality becomes clear if one realizes that under uncertainty dynamic efficiency is a very coarse efficiency benchmark. The reason for this can be seen by comparing dynamic efficiency and exchange efficiency under certainty to the case of uncertainty. Under certainty both dynamic efficiency and exchange efficiency reduce to the same criterion and therefore rule out Ponzi schemes. Therefore the efficiency characterizations coincide. Under uncertainty, dynamic efficiency essentially rules out a reduction in savings at one node and a (weak) increase in aggregate consumption in all successor nodes (not just along one path succeeding the node where savings were reduced). The crucial point is that an attempt to lower savings at *one* node will affect capital accumulation in all successor nodes, even those which lie on efficient paths when these paths are viewed in isolation. This means that the instrument of savings adjustments cannot be finely tuned in the sense that only one or some (inefficient) paths along a subtree can be adjusted. But this is necessary for exchange efficiency. Exchange efficiency rules out that there is even a single *path* in the tree along which a Ponzi scheme can be played. This highlights the difference between efficient capital accumulation and risk sharing. Efficient risk sharing rules out even sophisticated Ponzi schemes, i.e. schemes which roll over debt along *some* paths in a subtree without affecting other paths in the subtree.

Summing up, we have shown that sequentially complete markets do not rule out risk sharing even in the presumably empirically relevant case of dynamic efficiency. This result can be interpreted as justification for a well designed PAYG social security system as insurance against macroeconomic risks during old age.

7 A Sufficient Condition for Interim Pareto Optimality

As for the case under certainty (see Balasko and Shell (1980), Okuno and Zilcha (1980)), it is possible to derive a stronger sufficient condition for Pareto optimality. This stronger criterion has the advantage that it does not require curvature assumptions on preferences and technology. Furthermore, it is satisfied in economies with land.

Under certainty the criterion requires $\liminf_{t\to\infty} ||p_t|| = 0$, if the resources of the economy

⁶For the sake of a simpler presentation we assumed in the theorems that allocations are uniformly bounded away from zero. In fact, we only need the consumption to be uniformly bounded away from zero along the subtree at which the condition on the sum of weighted prices holds (see Chattopadhyay/Gottardi (1999), Theorem 2).

are bounded. What the criterion essentially achieves is making the economy "quasi-finite", by putting a low weight, in form of a low value, on the future (or at least parts of the future). This fact allows one to use again the simple revealed preference proof of the first welfare theorem from standard Arrow-Debreu-economies. Thus, with this assumption, the reason for the failure of the first welfare theorem in OLG models, namely that prices do not necessarily form a linear functional on the commodity space, ⁷ does not have severe consequences in this case. In particular, economies with land (see Rhee (1991), Homburg (1991)), or more generally, economies with non-negligible consumers (see Geanakoplos and Polemarchakis (1991)) display equilibrium prices satisfying this property. We now state the corresponding assumption under uncertainty and prove the first welfare theorem. The assumption is similar to one given by Demange (1998).⁸

$$\liminf_{t \to \infty} \sum_{\sigma \in \Sigma_t} p(\sigma) = 0.$$
(9)

Under the condition (9) and under standard assumptions on preferences and technology, i.e. under the assumptions made in the paper with exception of the curvature assumptions, we may state:

Proposition 22 Suppose preferences and production technology satisfy standard assumptions. If (9) holds at a competitive equilibrium, the equilibrium allocation is interim Pareto-efficient.

Proof. Suppose not. Then there exists an alternative feasible allocation (\hat{x}, \hat{k}) which interim Pareto improves in comparison with the competitive allocation (x^*, k^*) . Since the consumers maximized their utility given the budget constraint in the competitive equilibrium, we must have

$$p(\sigma) \cdot \hat{x}(\sigma;\sigma) + \sum_{\sigma' \in \sigma^+} p(\sigma') \cdot \hat{x}(\sigma';\sigma) \ge p(\sigma) \cdot x^*(\sigma;\sigma) + \sum_{\sigma' \in \sigma^+} p(\sigma') \cdot x^*(\sigma';\sigma), \quad (10)$$

with strict inequality for at least one $\sigma \in \Sigma$. Let $\tilde{\sigma} \in \Sigma$ denote such a node. Furthermore,

⁷This implies that the value of a certain commodity bundle is not necessarily finite. As a consequence of this OLG economies can exhibit a "lack of market clearing at infinity" (see for example Geanakoplos and Polemarchakis (1991))

 $^{^{8}}$ A related result in a model with exogenous stochastic development of wages and interest rates can be found in Richter (1993).

since firms maximize expected profits in the competitive equilibrium

$$\sum_{\sigma'\in\sigma^{+}} p\left(\sigma'\right) \cdot f\left(k^{*}\left(\sigma\right),\theta\right) - p\left(\sigma\right) \cdot k^{*}\left(\sigma\right) \ge \sum_{\sigma'\in\sigma^{+}} p\left(\sigma'\right) \cdot f\left(\widehat{k}\left(\sigma\right),\theta\right) - p\left(\sigma\right) \cdot \widehat{k}\left(\sigma\right).$$
(11)

Let $\varepsilon = p(\widetilde{\sigma}) \cdot \widehat{x}(\widetilde{\sigma}; \widetilde{\sigma}) + \sum_{\sigma' \in \widetilde{\sigma}^+} p(\sigma') \cdot \widehat{x}(\sigma'; \widetilde{\sigma}) - p(\widetilde{\sigma}) \cdot x^*(\widetilde{\sigma}; \widetilde{\sigma}) + \sum_{\sigma' \in \widetilde{\sigma}^+} p(\sigma') \cdot x^*(\sigma'; \widetilde{\sigma}) > 0.$ Let $\Gamma_{\widetilde{\sigma}}$ denote the subtree that contains all successor nodes of $\widetilde{\sigma}$. Now fix some $T > t(\widetilde{\sigma})$. Summing form 0 to T over all nodes in the tree Γ gives

$$\sum_{\sigma \in \Gamma \cap \left(\cup_{t \leq T} \Sigma_t \right)} p\left(\sigma\right) \cdot \widehat{x}\left(\sigma; \sigma\right) + \sum_{\sigma' \in \sigma^+} p\left(\sigma'\right) \cdot \widehat{x}\left(\sigma'; \sigma\right)$$
(12)

$$-\sum_{\sigma\in\Gamma\cap\left(\cup_{t\leq T-1}\Sigma_{t}\right)}\sum_{\sigma'\in\sigma^{+}}p\left(\sigma'\right)\cdot f\left(\widehat{k}\left(\sigma\right),\theta\right)+p\left(\sigma\right)\cdot\widehat{k}\left(\sigma\right)>$$

$$\sum_{\boldsymbol{\sigma}\in\Gamma\cap\left(\cup_{t\leq T}\Sigma_{t}\right)}p\left(\boldsymbol{\sigma}\right)\cdot\boldsymbol{x}^{*}\left(\boldsymbol{\sigma};\boldsymbol{\sigma}\right)+\sum_{\boldsymbol{\sigma}'\in\boldsymbol{\sigma}^{+}}p\left(\boldsymbol{\sigma}'\right)\cdot\boldsymbol{x}^{*}\left(\boldsymbol{\sigma}';\boldsymbol{\sigma}\right)-\sum_{\boldsymbol{\sigma}\in\Gamma\cap\left(\cup_{t\leq T-1}\Sigma_{t}\right)}\sum_{\boldsymbol{\sigma}'\in\boldsymbol{\sigma}^{+}}p\left(\boldsymbol{\sigma}'\right)\cdot\boldsymbol{f}\left(\boldsymbol{k}^{*}\left(\boldsymbol{\sigma}\right),\boldsymbol{\theta}\right)+p\left(\boldsymbol{\sigma}\right)\cdot\boldsymbol{k}^{*}\left(\boldsymbol{\sigma}\right),$$

where the difference between the left-hand and right hand expression is is at least $\varepsilon > 0$. Forming the same sum with the exception of not including the successor nodes of $\Sigma_{T_{\tau}}$ by feasibility of both allocations, taking differences we get

$$\sum_{\sigma \in \Sigma_{T+1}} p(\sigma) \cdot \widehat{x}(\sigma, \sigma_{-1}) > \sum_{\sigma \in \Sigma_{T+1}} p(\sigma) \cdot x^*(\sigma, \sigma_{-1}), \qquad (13)$$

with the difference again being at least ε . But by assumption (9) and the boundedness of the economy this is a contradiction.

The efficiency of an economy with land is a corollary of this result. We define land as an object which pays a fixed amount of the single commodity at each time in each state, which we denote by D. Note that this more restrictive than necessary and done for notational simplicity. Under certainty (see Rhee (1991), Homburg (1991)), it is only necessary that the income share of land does not vanish asymptotically. Given the boundedness of our economy it would therefore suffice to assume that the return from land is bounded away from zero across all nodes. In equilibrium, the value of land must be finite, i.e.

 $R \cdot \sum_{\sigma \in \Sigma} p(\sigma) < \infty$. This clearly requires (9). Thus we can state:

Corollary 23 A competitive equilibrium in an economy with land is always interim Pareto optimal.

Remark 24 It can also be shown that the sufficient condition for interim Pareto optimality stated in terms of a condition on the ratio of (net) dividends and the value of the market portfolio in Abel, Mankiw, Summers and Zeckhauser (1989) implies (9).

8 Conclusions

We have given a complete characterization of interim Pareto optimality in a stochastic OLG model with production and social security. We have shown that the risk sharing possibilities in a model with production do not improve compared to a pure exchange model, although the redistributional possibilities improve. Our first main result gives a characterization of interim Pareto optimality. In a competitive equilibrium, interim Pareto optimality is equivalent to intergenerational exchange efficiency, which in turn implies dynamic efficiency. Our characterization subsumes also equilibria with PAYG social security system. If the condition for exchange efficiency is violated, there is a Pareto-improving role for a PAYG social security system. Our second main result shows that dynamic efficiency does not rule out a Pareto-improving role for a social security system to act as insurance against aggregate productivity risk in the second period of life (old age) through dynamic risk sharing. Furthermore, this kind of risk sharing cannot be replicated on the capital market as it insures against aggregate (macroeconomic) risk. We have also provided a stronger sufficient criterion for interim Pareto optimality and have shown that economies with land satisfy this criterion. Moreover, we have proved a second welfare theorem for our economy.

It is important to note that the fact that there may be risk sharing possibilities in a competitive equilibrium of our economy are not derived from market incompleteness but from the dynamic structure of the economy. We have interpreted this kind of risk sharing as sophisticated Ponzi scheme as it involves collecting *contributions* from the young but rolls over an *insurance contract* rather than *debt* from generation to generation. The importance of our results stems from the fact that dynamic efficiency is an empirically testable criterion which implies that Ponzi schemes are infeasible. Therefore, if it holds, a wide range of Pareto-improving policies are ruled out. Our results, however, show that

this testable criterion is not sufficient to guarantee even the weak notion of interim Pareto optimality of a competitive market allocation.

Our results can be extended in several directions. First, it would be interesting to derive conditions for dynamic efficiency and Pareto optimality of competitive allocations that can easily be verified empirically as, for example, the sufficient condition for dynamic efficiency and interim Pareto optimality in Abel, Summers, Mankiw and Zeckhauser (1989). Second, one can hope to find simpler efficiency characterizations for stationary economies. This is at least true for pure exchange models and is likely to carry over to a model with capital (Demange and Laroque (2000)).

Appendix: Proof of Theorem 13

Proof. Clearly, if the above conditions holds, then the allocation is not exchange efficient and therefore not interim Pareto optimal. For the converse recall that the characterizing condition for dynamic efficiency was $\sup_{\sigma^{\infty} \in \Gamma_{\hat{\sigma}}} \sum_{t=\tau}^{\infty} \frac{1}{\psi(\sigma_t)}, \sigma_t = (\theta_0, ..., \theta_t)$ not being bounded in every subtree $\Gamma_{\hat{\sigma}}$. Since $p(\sigma_t) = \psi(\sigma_t) \cdot \prod_{i=0}^t q_i(\theta_i | \sigma_{i-1}) \leq \psi(\sigma_t)$ exchange efficiency implies dynamic efficiency. To see this consider every subtree and then choose the induced weight functions on it to be equal to the probability of reaching the node. The condition for dynamic efficiency follows from the previous remark and the fact that each node is reached with strictly positive probability. To prove the result of the theorem it therefore remains to be shown that *joint deviations* from the equilibrium allocation cannot be Pareto-improving. The proof is completed by the following three steps:

Step 1: We will show that it is never optimal to increase savings at any point of time at any node in the tree. Suppose there is an increase in savings at node $\sigma = (\theta_0, ..., \theta_t)$ (at time t) and node σ is among the first nodes (in terms of time) at which there is an increase in saving. Consider the individual born at this node with utility function $u_{\sigma}(x(\sigma))$. Then there must be a decrease in the youth consumption of this individual which is defined by the amount of increase in saving. So let $\hat{x}(\sigma; \sigma)$ denote the youth consumption in node σ after the increase in saving and let $x^*(\sigma; \sigma)$, as defined above, denote the equilibrium consumption when young. Since the new allocation is supposed to be Pareto-improving, we must have by the non-vanishing Gaussian curvature condition

$$\sum_{\sigma'\in\sigma^+} \delta_2\left(\sigma',\sigma\right) \ge -\delta_1\left(\sigma\right) + \underline{\rho}\frac{\left(\delta_1\left(\sigma\right)\right)^2}{p\left(\sigma\right)}.$$
(14)

Rewriting this using the relation between ψ_{t+1}, ψ_t and p we get

$$\sum_{\theta \in A_{t+1}} q_{t+1}\left(\theta \,|\, \sigma_t\right) \cdot \frac{\psi_t\left(\sigma\right)}{f'\left(k^*\left(\sigma\right), \theta\right)} \cdot \left[\widehat{x}\left(\sigma'; \sigma\right) - x^*\left(\sigma'; \sigma\right)\right] \ge$$
(15)

$$-\psi_{t}\left(\sigma\right)\cdot\left(\widehat{x}\left(\sigma;\sigma\right)-x^{*}\left(\sigma;\sigma\right)\right)+\underline{\rho}\frac{\left(\psi_{t}\left(\sigma\right)\cdot\left(\widehat{x}\left(\sigma;\sigma\right)-x^{*}\left(\sigma;\sigma\right)\right)\right)^{2}}{\psi_{t}\left(\sigma\right)}$$

Noting that $\hat{k}(\sigma) - k^*(\sigma) = -(\hat{x}(\sigma;\sigma) - x^*(\sigma;\sigma)) > 0$, replacing ψ_t with ψ_{t+1}/f' , neglecting the last term on the right and side and averaging with weights gives

$$\sum_{\theta \in A_{t+1}} q_{t+1}\left(\theta | \sigma_{t}\right) \cdot \psi_{t+1}\left(\sigma'\right) \cdot \left[\widehat{x}\left(\sigma';\sigma\right) - x^{*}\left(\sigma';\sigma\right)\right] >$$
(16)

$$\sum_{\theta \in A_{t+1}} q_{t+1}\left(\theta | \sigma_{t}\right) \cdot \psi_{t+1}\left(\sigma\right) \cdot f'\left(k^{*}\left(\sigma\right), \theta\right) \cdot \left[\widehat{k}\left(\sigma\right) - k^{*}\left(\sigma\right)\right],$$

i.e. that due to the strict concavity of the production function, the value of the necessary increase in tomrrow's consumption is strictly larger than the increase in tomorrow's production induced by the increase in saving.

This result shows that a Pareto-improving new allocation can never begin with an increase in saving.

Step 2: Suppose now $\sigma = (\theta_0, ..., \theta_t)$ (at time t) is among the first nodes in time at which there is a decrease in saving. Consider the individual born at this node with utility function $u_{\sigma}(x(\sigma))$. Let us, as in the preceding paragraph, denote the new allocation by a. So suppose in this case $\hat{k}(\sigma) - k^*(\sigma) = -(\hat{x}(\sigma; \sigma) - x^*(\sigma; \sigma)) < 0$. The argument is now similar to the one before. Again, by non-vanishing Gaussian curvature

$$\sum_{\sigma' \in \sigma^+} \delta_2\left(\sigma', \sigma\right) \ge -\delta_1\left(\sigma\right) + \underline{\rho} \frac{\left(\delta_1\left(\sigma\right)\right)^2}{p\left(\sigma\right)}.$$
(17)

Note that $\sum_{\sigma' \in \sigma^+} \delta_2(\sigma', \sigma) \leq 0$ and $-\delta_1(\sigma) < 0$ in this case, so that

$$-\sum_{\sigma'\in\sigma^+}\delta_2\left(\sigma',\sigma\right)<\delta_1\left(\sigma\right).$$
(18)

Writing out the expressions and substituting as in the argument before

$$\sum_{\theta \in A_{t+1}} q_{t+1}\left(\theta \mid \theta_{t}\right) \cdot \psi_{t+1}\left(\sigma'\right) \cdot \left[x^{*}\left(\sigma';\sigma\right) - \widehat{x}\left(\sigma';\sigma\right)\right] <$$
(19)

$$\sum_{\theta \in A_{t+1}} q_{t+1}\left(\theta \right| \theta_{t}\right) \cdot \psi_{t+1}\left(\sigma\right) \cdot f'\left(k^{*}\left(\sigma\right), \theta\right) \cdot \left[k^{*}\left(\sigma\right) - \widehat{k}\left(\sigma\right)\right],$$

i.e. the value of reduction in output at the successor nodes of σ is larger than the maximal possible reduction in value of consumption that leaves the individual indifferent.

These two facts together imply that if there is an interim Pareto-improving allocation it must be of the following form. At the first node where it differs from the initial allocation there is either a pure redistributive transfer (which necessarily involves a positive transfer for the old generation) or a decrease in saving which gives the amount less saved as consumption to the old generation , or there is a combination of both. This fact along with the quasi-concavity of utility functions and strict convexity of the technology imply that there will never be an increase in saving at any node in the tree.

Step 3: Assume a Pareto improvement is possible. Consider the following identity, which follows from the resource constraint $x(\sigma'; \sigma) + x(\sigma'; \sigma') + k(\sigma') = f(k(\sigma), \theta)$ $\forall \sigma' = (\theta, \sigma) \in \sigma^+$

$$\Delta x \left(\sigma'; \sigma\right) + \Delta x \left(\sigma'; \sigma'\right) + \Delta k \left(\sigma'\right) = \Delta f \left(k \left(\sigma\right), \theta\right), \tag{20}$$

where $\Delta x (\sigma'; \sigma) = (\hat{x} (\sigma'; \sigma) - x^* (\sigma'; \sigma))$ etc. if (x^*, k^*) is the initial competitive equilibrium allocation and (\hat{x}, \hat{k}) is the new interim Pareto-improving allocation. Equivalently

$$-\Delta x \left(\sigma'; \sigma'\right) - \Delta k \left(\sigma'\right) = \Delta x \left(\sigma'; \sigma\right) - \Delta f \left(k \left(\sigma\right), \theta\right).$$
(21)

If we define $\Delta \varepsilon (\sigma') = -\Delta k (\sigma')$ as the dissaving at node σ' when changing to the new allocation, which is by the argument made above always nonnegative, and $\Delta a (\sigma'; \sigma') = -\Delta x (\sigma'; \sigma')$ as the decrease in youth consumption when changing to the new allocation, we have $\Delta a (\sigma'; \sigma') + \Delta \varepsilon (\sigma') > 0^9$

$$\Delta a\left(\sigma';\sigma'\right) + \Delta \varepsilon\left(\sigma'\right) = \Delta x\left(\sigma';\sigma\right) - \Delta f\left(k\left(\sigma\right),\theta\right).$$
(22)

⁹It is well possible that there is an increase in youth consumption at some later node, however the sum $\Delta a (\sigma'; \sigma') + \Delta \varepsilon (\sigma')$ cannot become zero or negative, since such increases in youth consumption just cancel out with the decrease in saving.

Consider all nodes $\sigma' \in \sigma^+$ for which $\Delta a(\sigma'; \sigma') + \Delta \varepsilon(\sigma') > 0$. Since the improving allocation $(\widehat{x}, \widehat{k})$ must at some node $\widetilde{\sigma}$ be different from the initial one, by the arguments made above we must have $\Delta a(\widetilde{\sigma}; \widetilde{\sigma}) + \Delta \varepsilon(\widetilde{\sigma}) > 0$. Consider now the successor nodes of $\widetilde{\sigma}$ for which $\Delta a(\sigma"; \sigma") + \Delta \varepsilon(\sigma") > 0$, $\sigma" \in \widetilde{\sigma}^+$. It is easy to see that if we continue this way we inductively define a subtree, called $\Gamma_{\widetilde{\sigma}}$. Multiplying with contingent prices $p(\sigma')$ and summing over $\sigma' \in \sigma^+ \cap \Gamma_{\widetilde{\sigma}}$ gives

$$\sum_{\sigma' \in \sigma^{+} \cap \Gamma_{\tilde{\sigma}}} p(\sigma') \cdot \left[\Delta a(\sigma'; \sigma') + \Delta \varepsilon(\sigma') \right] = \sum_{\sigma' \in \sigma^{+} \cap \Gamma_{\tilde{\sigma}}} p(\sigma') \cdot \left[\Delta x(\sigma'; \sigma) - \Delta f(k(\sigma), \theta) \right].$$
(23)

Next, we state a technical lemma which follows from the elasticity assumption (4) and gives a quadratic term lower bound for changes in the value of production due to changes in investment behavior similar to the non-vanishing Gaussian curvature assumption for the utility functions. We omit a proof of this lemma. It follows from standard arguments (see Zilcha (1990)).

Lemma 25 Given assumption (4) on the production function and the boundedness of the economy, the following holds for all $0 < \delta < k(\sigma) < \sup_{\theta \in \cup_{t \ge 0} A_t} \{k | f(k, \theta) = k\}$ for all $\sigma \in \Sigma$ and a price system p consistent with $k(\sigma)$ as defined in (2)

$$\sum_{\sigma' \in \sigma^+} p\left(\sigma'\right) \cdot \left[f\left(k\left(\sigma\right), \theta_{t+1}\right) - f\left(k\left(\sigma\right) - \delta, \theta_{t+1}\right)\right] \ge p\left(\sigma\right) \cdot \delta + c \cdot \frac{\left(p\left(\sigma\right) \cdot \delta\right)^2}{p\left(\sigma\right)}.$$

Using now the non-vanishing Gaussian curvature condition for preferences and the lemma above for production functions, we get^{10}

$$\sum_{\sigma' \in \sigma^+ \cap \Gamma_{\widetilde{\sigma}}} p\left(\sigma'\right) \cdot \left[\Delta x\left(\sigma';\sigma\right) - \Delta f\left(k\left(\sigma\right),\theta\right)\right] \ge$$
(24)

$$-\delta_{1}(\sigma) + \underline{\rho} \frac{\left(\delta_{1}(\sigma)\right)^{2}}{p(\sigma)} + p(\sigma) \cdot \Delta \varepsilon(\sigma) + c \cdot \frac{\left(p(\sigma) \cdot \Delta \varepsilon(\sigma)\right)^{2}}{p(\sigma)},$$

where $\delta_1(\sigma)$ is defined as above.

¹⁰Note that the inequality in (24) holds if we sum the left-hand side over σ^+ and therefore also holds by summing over the nonnegative terms, what is done in (24).

Replacing $\Delta x (\sigma'; \sigma) - \Delta f (k (\sigma), \theta)$ by $\Delta a (\sigma'; \sigma') + \Delta \varepsilon (\sigma')$ gives

$$\sum_{\sigma' \in \sigma^{+} \cap \Gamma_{\bar{\sigma}}} p(\sigma') \cdot \left[\Delta a(\sigma'; \sigma') + \Delta \varepsilon(\sigma') \right] \geq -\delta_{1}(\sigma) + \underline{\rho} \frac{(\delta_{1}(\sigma))^{2}}{p(\sigma)} + p(\sigma) \cdot \Delta \varepsilon(\sigma) + c \cdot \frac{(p(\sigma) \cdot \Delta \varepsilon(\sigma))^{2}}{p(\sigma)}.$$
(25)

Like Chattopadhyay and Gottardi (1999) we define a function $\lambda_{\Gamma_{\widetilde{\sigma}}}: \Gamma_{\widetilde{\sigma}} \to [0, 1]$ by

$$\lambda_{\Gamma_{\tilde{\sigma}}}\left(\sigma'\right) = \frac{p\left(\sigma'\right) \cdot \left[\Delta a\left(\sigma';\sigma'\right) + \Delta\varepsilon\left(\sigma'\right)\right]}{\sum\limits_{\sigma'\in\sigma^{+}\cap\Gamma_{\tilde{\sigma}}} p\left(\sigma'\right) \cdot \left[\Delta a\left(\sigma';\sigma'\right) + \Delta\varepsilon\left(\sigma'\right)\right]}.$$
(26)

Note that given the way $\Gamma_{\tilde{\sigma}}$ is constructed, $\lambda_{\Gamma_{\tilde{\sigma}}}$ is well defined, strictly positive and satisfies $\sum_{\sigma' \in \sigma^+ \cap \Gamma_{\tilde{\sigma}}} \lambda_{\Gamma_{\tilde{\sigma}}} (\sigma') = 1.$

Now we consider an arbitrary path $\sigma^{\infty}(\Gamma_{\tilde{\sigma}})$ in the subtree. Define $\gamma = \min\{c, \underline{\rho}\}$. Equation (25) can now be written as

$$\frac{1}{\lambda_{\Gamma_{\tilde{\sigma}}}(\sigma')} \cdot p(\sigma') \cdot \left[\Delta a(\sigma'; \sigma') + \Delta \varepsilon(\sigma')\right] \ge$$
(27)

$$p(\sigma) \cdot \Delta a(\sigma; \sigma) + \gamma \frac{\left(p(\sigma) \cdot \Delta a(\sigma; \sigma)\right)^2}{p(\sigma)} + p(\sigma) \cdot \Delta \varepsilon(\sigma) + \gamma \frac{\left(p(\sigma) \cdot \Delta \varepsilon(\sigma)\right)^2}{p(\sigma)}.$$

Inverting both sides of this equation we obtain

$$\frac{\lambda_{\Gamma_{\tilde{\sigma}}}\left(\sigma'\right)}{p\left(\sigma'\right)\cdot\left[\Delta a\left(\sigma';\sigma'\right)+\Delta\varepsilon\left(\sigma'\right)\right]} \le$$
(28)

$$\frac{1}{p\left(\sigma\right)\cdot\Delta a\left(\sigma;\sigma\right)+\gamma\frac{\left(p(\sigma)\cdot\Delta a\left(\sigma;\sigma\right)\right)^{2}}{p(\sigma)}+p\left(\sigma\right)\cdot\Delta\varepsilon\left(\sigma\right)+\gamma\frac{\left(p(\sigma)\cdot\Delta\varepsilon\left(\sigma\right)\right)^{2}}{p(\sigma)}}$$

for all $\sigma \in \Gamma_{\widetilde{\sigma}}$.

This is equivalent to

$$\frac{\lambda_{\Gamma_{\tilde{\sigma}}}\left(\sigma'\right)}{p\left(\sigma'\right)\cdot\left[\Delta a\left(\sigma';\sigma'\right)+\Delta\varepsilon\left(\sigma'\right)\right]} \leq$$
(29)

$$\frac{1}{p(\sigma) \cdot [\Delta a(\sigma; \sigma) + \Delta \varepsilon(\sigma)]} \cdot \left(1 - \frac{\frac{\gamma}{p(\sigma)} \cdot [(p(\sigma) \cdot \Delta a(\sigma; \sigma))^2 + (p(\sigma) \cdot \Delta \varepsilon(\sigma))^2]}{p(\sigma) \cdot \Delta a(\sigma; \sigma) + \gamma \frac{(p(\sigma) \cdot \Delta a(\sigma; \sigma))^2}{p(\sigma)} + p(\sigma) \cdot \Delta \varepsilon(\sigma) + \gamma \frac{(p(\sigma) \cdot \Delta \varepsilon(\sigma))^2}{p(\sigma)}}\right).$$

Further algebraic manipulations on the right-hand side give

$$\frac{\lambda_{\Gamma_{\tilde{\sigma}}}\left(\sigma'\right)}{p\left(\sigma'\right)\cdot\left[\Delta a\left(\sigma';\sigma'\right)+\Delta\varepsilon\left(\sigma'\right)\right]} \leq$$
(30)

$$\frac{1}{p(\sigma) \cdot [\Delta a(\sigma; \sigma) + \Delta \varepsilon(\sigma)]} - \frac{1}{\frac{p(\sigma) \cdot [\Delta a(\sigma; \sigma) + \Delta \varepsilon(\sigma)])^2}{\frac{\gamma}{p(\sigma)} \cdot [(p(\sigma) \cdot \Delta a(\sigma; \sigma))^2 + (p(\sigma) \cdot \Delta \varepsilon(\sigma))^2]} + p(\sigma) \cdot [\Delta a(\sigma; \sigma) + \Delta \varepsilon(\sigma)]}$$

so that we finally obtain

$$\frac{\lambda_{\Gamma_{\widetilde{\sigma}}}\left(\sigma'\right)}{p\left(\sigma'\right)\cdot\left[\Delta a\left(\sigma';\sigma'\right)+\Delta\varepsilon\left(\sigma'\right)\right]} \leq$$
(31)

$$\frac{1}{p\left(\sigma\right)\cdot\left[\Delta a\left(\sigma;\sigma\right)+\Delta\varepsilon\left(\sigma\right)\right]}-\frac{\gamma}{p\left(\sigma\right)}\frac{1}{\frac{\left[\Delta a\left(\sigma;\sigma\right)+\Delta\varepsilon\left(\sigma\right)\right]^{2}}{\left(\Delta a\left(\sigma;\sigma\right)\right)^{2}+\left(\Delta\varepsilon\left(\sigma\right)\right)^{2}}+\Delta a\left(\sigma;\sigma\right)+\Delta\varepsilon\left(\sigma\right)}$$

We want to show next that the expression $\frac{[\Delta a(\sigma;\sigma) + \Delta \varepsilon(\sigma)]^2}{(\Delta a(\sigma;\sigma))^2 + (\Delta \varepsilon(\sigma))^2} + \Delta a(\sigma;\sigma) + \Delta \varepsilon(\sigma)$, which by the assumptions made on the subtree $\Gamma_{\tilde{\sigma}}$ strictly positive, is also bounded above. By the resource constraint and the assumption that the economy is bounded, $\Delta a(\sigma;\sigma) + \Delta \varepsilon(\sigma)$ is clearly bounded above. Rewrite $\frac{[\Delta a(\sigma;\sigma) + \Delta \varepsilon(\sigma)]^2}{(\Delta a(\sigma;\sigma))^2 + (\Delta \varepsilon(\sigma))^2}$ as

$$\frac{\left(\Delta a\left(\sigma;\sigma\right)\right)^{2}}{\left(\Delta a\left(\sigma;\sigma\right)\right)^{2} + \left(\Delta\varepsilon\left(\sigma\right)\right)^{2}} + \frac{2\Delta a\left(\sigma;\sigma\right)\Delta\varepsilon\left(\sigma\right)}{\left(\Delta a\left(\sigma;\sigma\right)\right)^{2} + \left(\Delta\varepsilon\left(\sigma\right)\right)^{2}} + \frac{\left(\Delta\varepsilon\left(\sigma\right)\right)^{2}}{\left(\Delta a\left(\sigma;\sigma\right)\right)^{2} + \left(\Delta\varepsilon\left(\sigma\right)\right)^{2}}.$$
 (32)

The first and the third term are bounded above by 1 and so is the middle term because

$$\frac{2\Delta a\left(\sigma;\sigma\right)\Delta\varepsilon\left(\sigma\right)}{\left(\Delta a\left(\sigma;\sigma\right)\right)^{2} + \left(\Delta\varepsilon\left(\sigma\right)\right)^{2}} = \frac{2}{\frac{\Delta a\left(\sigma;\sigma\right)}{\Delta\varepsilon\left(\sigma\right)} + \frac{\Delta\varepsilon\left(\sigma\right)}{\Delta a\left(\sigma;\sigma\right)}}$$
(33)

and the function $x + \frac{1}{x}$ is bounded below on the positive real line.

So there is a constant K > 0 such that $\frac{[\Delta a(\sigma;\sigma) + \Delta \varepsilon(\sigma)]^2}{(\Delta a(\sigma;\sigma))^2 + (\Delta \varepsilon(\sigma))^2} + \Delta a(\sigma;\sigma) + \Delta \varepsilon(\sigma) \leq K$. Inserting this into (31) gives

$$\frac{\lambda_{\Gamma_{\tilde{\sigma}}}(\sigma')}{p(\sigma') \cdot \left[\Delta a(\sigma';\sigma') + \Delta\varepsilon(\sigma')\right]} \le \frac{1}{p(\sigma) \cdot \left[\Delta a(\sigma;\sigma) + \Delta\varepsilon(\sigma)\right]} - \frac{\gamma}{p(\sigma)} \frac{1}{K}.$$
 (34)

If we now iterate this inequality along $\sigma^{\infty}(\Gamma_{\tilde{\sigma}})$ starting with $\tilde{\sigma}$, we obtain

$$\frac{\lambda_{\Gamma_{\widetilde{\sigma}}}\left(\sigma_{T}^{\infty}\right)...\lambda_{\Gamma_{\widetilde{\sigma}}}\left(\sigma_{t(\widetilde{\sigma})}^{\infty}\right)}{p\left(\sigma_{T}^{\infty}\right)\cdot\left[\Delta a\left(\sigma_{T}^{\infty};\sigma_{T}^{\infty}\right)+\Delta\varepsilon\left(\sigma_{T}^{\infty}\right)\right]}+\frac{\gamma}{K}\sum_{t=t(\widetilde{\sigma})}^{T-1}\frac{\lambda_{\Gamma_{\widetilde{\sigma}}}\left(\sigma_{t}^{\infty}\right)...\lambda_{\Gamma_{\widetilde{\sigma}}}\left(\sigma_{t(\widetilde{\sigma})}^{\infty}\right)}{p\left(\sigma_{t}^{\infty}\right)} \leq \frac{1}{p\left(\widetilde{\sigma}\right)\cdot\left[\Delta a\left(\widetilde{\sigma};\widetilde{\sigma}\right)+\Delta\varepsilon\left(\widetilde{\sigma}\right)\right]},$$
(35)

so that $\lim_{T\to\infty} \sum_{t=t(\tilde{\sigma})}^{T-1} \frac{\lambda_{\Gamma_{\tilde{\sigma}}}(\sigma_t^{\infty})...\lambda_{\Gamma_{\tilde{\sigma}}}(\sigma_{t(\tilde{\sigma})}^{\infty})}{p(\sigma_t^{\infty})}$ as being increasing in T must converge to a positive real number, call it A. So in the case that an interim Pareto improvement were possible, a subtree $\Gamma_{\tilde{\sigma}}$, a weight function $\lambda_{\Gamma_{\tilde{\sigma}}}$ and a finite number A would exist such that (5) would hold. This completes the proof.

References

- Abel, Andrew B., N. Gregory Mankiw, Lawrence Summers and Richard J. Zeckhauser (1989), Assessing Dynamic Efficiency: Theory and Evidence, Review of Economic Studies 56, 1-19.
- [2] Aiyagari, S. Rao and Dan Peled (1991), Dominant Root Characterization of Pareto Optimality and the Existence of Optimal Equilibria in Stochastic Overlapping Generations Models, Journal of Economic Theory 54, 69-83.
- [3] Aliprantis, Charalambos and Border, Kim (1994), Infinite Dimensional Analysis, Berlin et. al.: Springer-Verlag.
- [4] Balasko, Yves and Karl Shell (1980), The Overlapping-Generations Model, I: The Case of Pure Exchange without Money, Journal of Economic Theory 23, 281-306.
- [5] Balasko, Yves and Karl Shell (1981), The Overlapping-Generations Model. II: The Case of Pure Exchange with Money, Journal of Economic Theory 24, 112-42.
- [6] Benveniste, Lawrence and David Gale (1975), An Extension of Cass' Characterization of Infinite Efficient Production Programs, Journal of Economic Theory 10, 229-238.

- [7] Bose, Amitava (1974), Pareto optimality and efficient capital accumulation, Discussion Paper No. 74-4, Dept. of Economics, University of Rochester.
- [8] Bose, Amitava and Debraj Ray (1993), Monetary equilibrium in an overlapping generations model with productive capital, Economic Theory 3, 697-716.
- Breyer, Friedrich (1989), On the Intergenerational Pareto Efficiency of Pay-as-You-Go Financed Pension Systems, Journal of Institutional and Theoretical Economics 145, 643-58.
- [10] Cass, David (1972), On capital overaccumulation in the aggregative, neoclassical model of economic growth: A complete characterization, Journal of Economic Theory 4, 200-223.
- [11] Chattopadhyay, Subir and Piero Gottardi (1999), Stochastic OLG Models, Market Structure, and Optimality, Journal of Economic Theory 89, 21-67.
- [12] Dechert, W. Davis and Kenji Yamamoto (1992), Asset Valuation and Production Efficiency in an Overlapping-Generations Model with Production Shocks, Review of Economic Studies 59, 389-405.
- [13] Demange, Gabrielle (1998), On the efficiency of intergenerational risk sharing and capital accumulation in an economy with land, DELTA document no 98-21.
- [14] Demange, Gabrielle and Guy Laroque (1999), Social Security and Demographic Shocks, Econometrica 67, 527-42.
- [15] Demange, Gabrielle and Guy Laroque (2000), Social Security, Optimality, and Equilibria in a Stochastic Overlapping Generations Economy, Journal of Public Economic Theory 2, 1-23
- [16] Diamond, Peter (1965), National Debt in a Neoclassical Growth Model, American Economic Review 55, 1127-1155.
- [17] Dutta, Jayasri and Herakles Polemarchakis (1990), Asset Markets and Equilibrium Processes, Review of Economic Studies 57, 229-254.
- [18] Geanakoplos, John D. and Heraklis M. Polemarchakis (1991), Overlapping Generations, in: Hildenbrand, Werner and Hugo Sonnenschein (eds.), Handbook of mathematical economics. Volume 4. Elsevier Science, New York, 1899-1960.
- [19] Homburg, Stefan (1991), Interest and Growth in an Economy with Land, Canadian Journal of Economics 24, 450-59.

- [20] Manuelli, Rodolfo (1990), Existence and Optimality of Currency Equilibrium in Stochastic Overlapping Generations Models: The Pure Endowment Case, Journal of Economic Theory 51, 268-94.
- [21] Mitra, Tapan (1979), Identifying Inefficiency in Smooth Aggregative Models of Economic Growth: A Unifying Criterion, Journal of Mathematical Economics 6, 85-111.
- [22] Muench, Thomas (1977), Optimality, the Interaction of Spot and Futures Markets, and the Nonneutrality of Money in the Lucas Model, Journal of Economic Theory 15, 325-344.
- [23] Okuno, Masahiro and Itzhak Zilcha (1980), On the Efficiency of a Competitive Equilibrium in Infinite Horizon Monetary Economies, Review of Economic Studies 47, 797-807.
- [24] Peled, Dan (1982), Informational Diversity over Time and the Optimality of Monetary Equilibria, Journal of Economic Theory 28, 255-74.
- [25] Rhee, Changyong (1991), Dynamic Inefficiency in an Economy with Land, Review of Economic Studies 58, 791-97.
- [26] Richter, Wolfram (1993), Intergenerational Risk Sharing and Social Security in an Economy with Land, Journal of Economics, Suppl. 7, 91-103.
- [27] Samuelson, Paul A. (1958), An exact consumption-loan model of interest with or without the contrivance of money, Journal of Political Economy 66, 467-482.
- [28] Tirole, Jean (1985), Asset Bubbles and Overlapping Generations, Econometrica 53, 1499-1528.
- [29] Zilcha, Itzhak (1990), Dynamic Efficiency in Overlapping Generations Models with Stochastic Production, Journal of Economic Theory 52, 364-79.

IZA Discussion Papers

| No | Author(s) | Titel | Area | Date |
|-----|---------------------------------------|---|------|------|
| 111 | V. Sorm K. Terrell | Sectoral Restructuring and Labor Mobility: A Comparative Look at the Czech Republic | 1/4 | 2/00 |
| 112 | L. Bellmann T. Schank | Innovations, Wages and Demand for Heterogeneous Labour: New Evidence from a Matched Employer-Employee Data-Set | 5 | 2/00 |
| 113 | R. Euwals | Do Mandatory Pensions Decrease Household Savings? Evidence for the Netherlands | 3 | 2/00 |
| 114 | G. Brunello A. Medio | An Explanation of International Differences in Education and Workplace Training | 2 | 2/00 |
| 115 | A. Cigno F. C. Rosati | Why do Indian Children Work, and is it Bad for Them? | 3 | 2/00 |
| 116 | C. Belzil | Unemployment Insurance and Subsequent Job Duration: Job Matching vs. Unobserved Heterogeneity | 3 | 2/00 |
| 117 | S. Bender A. Haas C. Klose | IAB Employment Subsample 1975-1995. Opportunities for Analysis Provided by the Anonymised Subsample | 7 | 2/00 |
| 118 | M. A. Shields M. E. Ward | Improving Nurse Retention in the British National Health Service: The Impact of Job Satisfaction on Intentions to Quit | 5 | 2/00 |
| 119 | A. Lindbeck D. J. Snower | The Division of Labor and the Market for Organizations | 5 | 2/00 |
| 120 | P. T. Pereira P. S. Martins | Does Education Reduce Wage Inequality? Quantile Regressions Evidence from Fifteen European Countries | 5 | 2/00 |
| 121 | J. C. van Ours | Do Active Labor Market Policies Help Unemployed Workers to Find and Keep Regular Jobs? | 4/6 | 3/00 |
| 122 | D. Munich J. Svejnar K. Terrell | Returns to Human Capital under the Communist Wage Grid and During the Transition to a Market Economy | 4 | 3/00 |
| 123 | J. Hunt | Why Do People Still Live in East Germany? | 1 | 3/00 |
| 124 | R. T. Riphahn | Rational Poverty or Poor Rationality? The Take-up of Social Assistance Benefits | 3 | 3/00 |
| 125 | F. Büchel J. R. Frick | The Income Portfolio of Immigrants in Germany - Effects of Ethnic Origin and Assimilation. Or: Who Gains from Income Re-Distribution? | 1/3 | 3/00 |
| 126 | J. Fersterer R. Winter-Ebmer | Smoking, Discount Rates, and Returns to Education | 5 | 3/00 |

| 127 | M. Karanassou D. J. Snower | Characteristics of Unemployment Dynamics: The Chain Reaction Approach | 3 | 3/00 |
|-----|---|--|-----|------|
| 128 | O. Ashenfelter D. Ashmore O. Deschênes | Do Unemployment Insurance Recipients Actively Seek Work? Evidence From Randomized Trials in Four U.S. States | 6 | 3/00 |
| 129 | B. R. Chiswick M. E. Hurst | The Employment, Unemployment and Unemployment Compensation Benefits of Immigrants | 1/3 | 3/00 |
| 130 | G. Brunello S. Comi C. Lucifora | The Returns to Education in Italy: A New Look at the Evidence | 5 | 3/00 |
| 131 | B. R. Chiswick | Are Immigrants Favorably Self-Selected? An Economic Analysis | 1 | 3/00 |
| 132 | R. A. Hart | Hours and Wages in the Depression: British Engineering, 1926-1938 | 7 | 3/00 |
| 133 | D. N. F. Bell R. A. Hart O. Hübler W. Schwerdt | Paid and Unpaid Overtime Working in Germany and the UK | 1 | 3/00 |
| 134 | A. D. Kugler G. Saint-Paul | Hiring and Firing Costs, Adverse Selection and Long-term Unemployment | 3 | 3/00 |
| 135 | A. Barrett P. J. O'Connell | Is There a Wage Premium for Returning Irish Migrants? | 1 | 3/00 |
| 136 | M. Bräuninger M. Pannenberg | Unemployment and Productivity Growth: An Empirical Analysis within the Augmented Solow Model | 3 | 3/00 |
| 137 | JSt. Pischke | Continuous Training in Germany | 5 | 3/00 |
| 138 | J. Zweimüller R. Winter-Ebmer | Firm-specific Training: Consequences for Job Mobility | 1 | 3/00 |
| 139 | R. A. Hart Y. Ma | Wages, Hours and Human Capital over the Life Cycle | 1 | 3/00 |
| 140 | G. Brunello S. Comi | Education and Earnings Growth: Evidence from 11 European Countries | 2/5 | 4/00 |
| 141 | R. Hujer M. Wellner | The Effects of Public Sector Sponsored Training on Individual Employment Performance in East Germany | 6 | 4/00 |
| 142 | J. J. Dolado F. Felgueroso J. F. Jimeno | Explaining Youth Labor Market Problems in Spain: Crowding-Out, Institutions, or Technology Shifts? | 3 | 4/00 |
| 143 | P. J. Luke M. E. Schaffer | Wage Determination in Russia: An Econometric Investigation | 4 | 4/00 |
| 144 | G. Saint-Paul | Flexibility vs. Rigidity: Does Spain have the worst of both Worlds? | 1 | 4/00 |

| 145 | MS. Yun | Decomposition Analysis for a Binary Choice Model | 7 | 4/00 |
|-----|---|---|-----|------|
| 146 | T. K. Bauer J. P. Haisken-DeNew | Employer Learning and the Returns to Schooling | 5 | 4/00 |
| 147 | M. Belot J. C. van Ours | Does the Recent Success of Some OECD Countries in Lowering their Unemployment Rates Lie in the Clever Design of their Labour Market Reforms? | 3 | 4/00 |
| 148 | L. Goerke | Employment Effects of Labour Taxation in an Efficiency Wage Model with Alternative Budget Constraints and Time Horizons | 3 | 5/00 |
| 149 | R. Lalive J. C. van Ours J. Zweimüller | The Impact of Active Labor Market Programs and Benefit Entitlement Rules on the Duration of Unemployment | 3/6 | 5/00 |
| 150 | J. DiNardo K. F. Hallock JSt. Pischke | Unions and the Labor Market for Managers | 7 | 5/00 |
| 151 | M. Ward | Gender, Salary and Promotion in the Academic Profession | 5 | 5/00 |
| 152 | J. J. Dolado F. Felgueroso J. F. Jimeno | The Role of the Minimum Wage in the Welfare State: An Appraisal | 3 | 5/00 |
| 153 | A. S. Kalwij M. Gregory | Overtime Hours in Great Britain over the Period 1975-1999: A Panel Data Analysis | 3 | 5/00 |
| 154 | M. Gerfin M. Lechner | Microeconometric Evaluation of the Active Labour Market Policy in Switzerland | 6 | 5/00 |
| 155 | J. Hansen | The Duration of Immigrants' Unemployment Spells: Evidence from Sweden | 1/3 | 5/00 |
| 156 | C. Dustmann F. Fabbri | Language Proficiency and Labour Market Per- formance of Immigrants in the UK | 1 | 5/00 |
| 157 | P. Apps R. Rees | Household Production, Full Consumption and the Costs of Children | 7 | 5/00 |
| 158 | A. Björklund T. Eriksson M. Jäntti O. Raaum E. Österbacka | Brother Correlations in Earnings in Denmark, Finland, Norway and Sweden Compared to the United States | 5 | 5/00 |
| 159 | P J. Jost M. Kräkel | Preemptive Behavior in Sequential Tournaments | 5 | 5/00 |
| 160 | M. Lofstrom | A Comparison of the Human Capital and Signaling Models: The Case of the Self-Employed and the Increase in the Schooling Premium in the 1980's | 5 | 6/00 |
| 161 | V. Gimpelson D. Treisman G. Monusova | Public Employment and Redistributive Politics: Evidence from Russia's Regions | 4 | 6/00 |

| 162 | C. Dustmann M. E. Rochina- Barrachina | Selection Correction in Panel Data Models: An Application to Labour Supply and Wages | 6 | 6/00 |
|-----|---|--|---|------|
| 163 | R. A. Hart Y. Ma | Why do Firms Pay an Overtime Premium? | 5 | 6/00 |
| 164 | M. A. Shields S. Wheatley Price | Racial Harassment, Job Satisfaction and Intentions to Quit: Evidence from the British Nursing Profession | 5 | 6/00 |
| 165 | P. J. Pedersen | Immigration in a High Unemployment Economy: The Recent Danish Experience | 1 | 6/00 |
| 166 | Z. MacDonald M. A. Shields | The Impact of Alcohol Consumption on Occupa- tional Attainment in England | 5 | 6/00 |
| 167 | A. Barrett J. FitzGerald B. Nolan | Earnings Inequality, Returns to Education and Immigration into Ireland | 5 | 6/00 |
| 168 | G. S. Epstein A. L. Hillman | Social Harmony at the Boundaries of the Welfare State: Immigrants and Social Transfers | 3 | 6/00 |
| 169 | R. Winkelmann | Immigration Policies and their Impact: The Case of New Zealand and Australia | 1 | 7/00 |
| 170 | T. K. Bauer K. F. Zimmermann | Immigration Policy in Integrated National Economies | 1 | 7/00 |
| 171 | C. Dustmann F. Windmeijer | Wages and the Demand for Health – A Life Cycle Analysis | 5 | 7/00 |
| 172 | D. Card | Reforming the Financial Incentives of the Welfare System | 3 | 7/00 |
| 173 | D. S. Hamermesh | Timing, Togetherness and Time Windfalls | 5 | 7/00 |
| 174 | E. Fehr JR. Tyran | Does Money Illusion Matter? An Experimental Approach | 7 | 7/00 |
| 175 | M. Lofstrom | Self-Employment and Earnings among High- Skilled Immigrants in the United States | 1 | 7/00 |
| 176 | O. Hübler W. Meyer | Industrial Relations and the Wage Differentials between Skilled and Unskilled Blue-Collar Workers within Establishments: An Empirical Analysis with Data of Manufacturing Firms | 5 | 7/00 |
| 177 | B. R. Chiswick G. Repetto | Immigrant Adjustment in Israel: Literacy and Fluency in Hebrew and Earnings | 1 | 7/00 |
| 178 | R. Euwals M. Ward | The Renumeration of British Academics | 5 | 7/00 |
| 179 | E. Wasmer P. Weil | The Macroeconomics of Labor and Credit Market Imperfections | 2 | 8/00 |
| 180 | T. K. Bauer I. N. Gang | Sibling Rivalry in Educational Attainment: The German Case | 5 | 8/00 |
| | | | | |

| 181 | E. Wasmer Y. Zenou | Space, Search and Efficiency | 2 | 8/00 |
|-----|--|---|---|------|
| 182 | M. Fertig C. M. Schmidt | Discretionary Measures of Active Labor Market Policy: The German Employment Promotion Reform in Perspective | 6 | 8/00 |
| 183 | M. Fertig C. M. Schmidt | Aggregate-Level Migration Studies as a Tool for Forecasting Future Migration Streams | 1 | 8/00 |
| 184 | M. Corak B. Gustafsson T. Österberg | Intergenerational Influences on the Receipt of Unemployment Insurance in Canada and Sweden | 3 | 8/00 |
| 185 | H. Bonin K. F. Zimmermann | The Post-Unification German Labor Market | 4 | 8/00 |
| 186 | C. Dustmann | Temporary Migration and Economic Assimilation | 1 | 8/00 |
| 187 | T. K. Bauer M. Lofstrom K. F. Zimmermann | Immigration Policy, Assimilation of Immigrants and Natives' Sentiments towards Immigrants: Evidence from 12 OECD-Countries | 1 | 8/00 |
| 188 | A. Kapteyn A. S. Kalwij A. Zaidi | The Myth of Worksharing | 5 | 8/00 |
| 189 | W. Arulampalam | Is Unemployment Really Scarring? Effects of Unemployment Experiences on Wages | 3 | 8/00 |
| 190 | C. Dustmann I. Preston | Racial and Economic Factors in Attitudes to Immigration | 1 | 8/00 |
| 191 | G. C. Giannelli C. Monfardini | Joint Decisions on Household Membership and Human Capital Accumulation of Youths: The role of expected earnings and local markets | 5 | 8/00 |
| 192 | G. Brunello | Absolute Risk Aversion and the Returns to Education | 5 | 8/00 |
| 193 | A. Kunze | The Determination of Wages and the Gender Wage Gap: A Survey | 5 | 8/00 |
| 194 | A. Newell F. Pastore | Regional Unemployment and Industrial Restructuring in Poland | 4 | 8/00 |
| 195 | F. Büchel A. Mertens | Overeducation, Undereducation, and the Theory of Career Mobility | 5 | 9/00 |
| 196 | J. S. Earle K. Z. Sabirianova | Equilibrium Wage Arrears: A Theoretical and Empirical Analysis of Institutional Lock-In | 4 | 9/00 |
| 197 | G. A. Pfann | Options to Quit | 1 | 9/00 |

| 198 | M. Kreyenfeld C. K. Spiess G. G. Wagner | A Forgotten Issue: Distributional Effects of Day Care Subsidies in Germany | 3 | 9/00 |
|-----|---|--|---|-------|
| 199 | H. Entorf | Rational Migration Policy Should Tolerate Non- Zero Illegal Migration Flows: Lessons from Modelling the Market for Illegal Migration | 1 | 9/00 |
| 200 | T. Bauer G. S. Epstein I. N. Gang | What are Migration Networks? | 1 | 9/00 |
| 201 | T. J. Dohmen G. A. Pfann | Worker Separations in a Nonstationary Corporate Environment | 1 | 9/00 |
| 202 | P. Francois J. C. van Ours | Gender Wage Differentials in a Competitive Labor Market: The Household Interaction Effect | 5 | 9/00 |
| 203 | J. M. Abowd F. Kramarz D. N. Margolis T. Philippon | The Tail of Two Countries: Minimum Wages and Employment in France and the United States | 5 | 9/00 |
| 204 | G. S. Epstein | Labor Market Interactions Between Legal and Illegal Immigrants | 1 | 10/00 |
| 205 | A. L. Booth M. Francesconi J. Frank | Temporary Jobs: Stepping Stones or Dead Ends? | 1 | 10/00 |
| 206 | C. M. Schmidt R. Baltussen R. Sauerborn | The Evaluation of Community-Based Inter- ventions: Group-Randomization, Limits and Alternatives | 6 | 10/00 |
| 207 | C. M. Schmidt | Arbeitsmarktpolitische Maßnahmen und ihre Evaluierung: eine Bestandsaufnahme | 6 | 10/00 |
| 208 | J. Hartog R. Winkelmann | Dutch Migrants in New Zealand: Did they Fare Well? | 1 | 10/00 |
| 209 | M. Barbie M. Hagedorn A. Kaul | Dynamic Efficiency and Pareto Optimality in a Stochastic OLG Model with Production and Social Security | 3 | 10/00 |